Licensing Act 2003

Draft Southwark Statement of Licensing Policy 2011-14

LEGAL BACKGROUND TO THIS DOCUMENT

The Licensing Act 2003 came into effect on 24 November 2005. The Act repealed existing legislation relating to alcohol, entertainment and late night refreshment and introduced a new licensing regime, administered by the local licensing authority.

Section 5 of the Licensing Act 2003 requires each licensing authority to prepare and publish a statement of its licensing policy every three years. The initial Southwark Statement of Licensing Policy received council assembly approval on 8 December 2004. It was developed through public consultation and with full regard to the Act; the secondary regulations made under the Act; and the guidance produced by the Secretary of State for the Department of Culture Media and Sport (DCMS). The council's licensing policy has since been frequently updated and amended, in order to keep abreast of changes in law, guidance and local policy. Such amendments have been formally approved in January 2007, April and November 2008 and November 2009.

This document provides the fifth revision of the statement of policy and represents the authority's policy for the period 2011 - 2014. Regard has been had during the preparation of this latest revision to section 182 Guidance, issued by the Home Office in October 2010.

Revisions to the guidance are issued by the Secretary of State from time to time. When revisions are issued, there may be periods of time where the Southwark policy is inconsistent with the guidance. During such periods this council will have regard, and give appropriate weight, to the revised guidance.

It is important to note that a licensing authority may depart from its own policy if the individual circumstances of any case merit such a decision in the interests of the promotion of the licensing objectives.

CONTENTS

	Page No
Section 1 – Executive summary	5
Section 2 – All about Southwark	7
- Our borough	8
- Leading Southwark	9
Section 3 – Purpose and scope of the licensing policy	11
 Purpose of the policy 	11
- Scope of the policy	11
- Licence types	11
- Definitions	12
- The licensing objectives	12
- General information	13
Section 4 – Administration, exercise and delegation of function	15
- Applications for personal licences	15
- Applications for premises licences	15
a) Provision of scale plans	15
b) Variations and new applications	16
c) Provisional statements	16
- Applications for club premises certificates	16
- Temporary event notices	17
- Major festivals	19
- Consultation arrangements	19
- Responsible authorities	20
- Interested parties	20
- Representations	21
- Determination of applications	23
- Licence reviews	23
- 2012 Olympics	23
Section 5 – Determining Applications for Premises Licences and Premises Certificates	
	24 24
- How this policy works	2 4 25
- Conditions	25 26
 Duplication Section 6 – Other policies, objectives and guidance 	20 27
- Other statutory requirements	27
- Other statutory requirements - Other strategies and policies	29
- Guidance documents	31
Section 7 – The first licensing objective – The prevention of crim	
Section 7 - The mist incensing objective - The prevention of crim	32
- General	32
- Mandatory conditions	33
- Cumulative effect	37
- Saturation policies	38
- Camberwell saturation policy	39
- Peckham saturation policy	39
- Borough and Bankside policy	39
- Licensing hours	40

-	Designated premises supervisors and personal licence holders	40
-	Door Supervision	41
-	Police promotion event risk-assessment form 696	42
-	Entertainment involving striptease and nudity	42
-	The sale and possession of contraband and counterfeit goods	45
Secti	on 8 – The second licensing objective – Ensuring public safety	46
-	General	46
-	Other relevant legislation	47
-	A broad guide to risk assessments	48
-	Safe capacities	51
-	A proactive approach to customer safety	52
Secti	on 9 – The third licensing objective – The prevention of nuisance	53
-	General	53
-	Outdoor events	58
-	Nuisance outside of premises	59
-	Smoking and licensed premises	59
Secti	on 10 - The fourth licensing objective - The protection of children from	n
harm		61
-	General	61
-	Access to licensed premises	62
-	The purchase and consumption of alcohol by minors	63
-	Children and responsible drinks promotions	65
-	Children and cinemas	66
-	Children and regulated entertainments	66
Secti	on 11 – Enforcement	69
-	General	69
Secti	on 12 - Contact details	71
-	The licensing service	71
-	Other responsible authorities	71
-	Other agencies	73
	ŭ	
Diag	rams and tables	
	of Southwark	7
	e 1 – Table of delegations of licensing functions	22
	2 – Broad guide to the range of crime and disorder controls that should be	
	dered for various categories of premises operation	34
	- Key to table 1	35
•	e 3 – Broad guide to noise controls	52
	e 4 – Guide to proof of age controls	63

SECTION ONE - EXECUTIVE SUMMARY

Our policy recognises the change that is taking place within Southwark and how recent development and regeneration projects are adding to the growing value and importance of our borough. It recognises the extent of the contribution of the leisure and entertainments industry.

Southwark enjoys a widespread and diverse selection of licensed premises and venues. More than 1,200 premises are currently licensed for either the sale or supply of alcohol; the provision of regulated entertainment; and / or the provision of late night refreshment. These range from off-licences, shops and supermarkets; restaurants, cafes and take-away establishments; to pubs, bars, members clubs, night clubs, theatres, cinemas and indoor sports facilities. Together they combine to provide a wide-range of leisure and cultural opportunities; support tourism; provide employment; and make a significant economic contribution to the local community.

As long as premises management strive to act responsibly; run safe, well managed venues and facilities; and work together with the local community, they can make a positive contribution toward building community cohesion and cultural development.

Of course, negative impacts can also occur if good management practices are not followed. Potential negative impacts may arise in the form of noise, nuisance, disturbance and crime and disorder problems. We recognise that the misuse of alcohol particularly, while maintaining an important position in our leisure activity, does negatively impact upon both public health and the public purse through the demands made upon A&E; additional policing; additional street cleaning; and the criminal justice system.

Our policy, therefore, seeks to provide a necessary balance between providing a platform upon which responsible business operators may contribute towards a thriving business and late night economy while ensuring that the quality of life of those who live and work in the borough is protected and enhanced through the licensing system. We believe these aims are achievable if all parties concerned work together.

The opening sections of this policy help establish how we address our licensing responsibilities under the Act. Section two of our policy provides an introduction to our borough and our council. Section three sets out the scope of the policy and section four our administrative arrangements. Section five gives an explanation to the determinations process under the Act and section six sets out the background of other legislation, policies, objectives and guidance which support this policy.

Perhaps some of the most important matter is contained in sections seven through to ten in which we cover each of the four licensing objectives in detail. Here we offer guidance on the considerations we would like to see applicants for licences taking into account when conducting risk-assessments of their proposed operation in advance of compiling management operation plans and operating schedules.

Within these, section seven deals with the prevention of crime and disorder. It includes explanation of how this council monitors the impact of its licensing decisions through regular statistical analysis. It details local saturation policies that have been put in place where there is evidence of detrimental cumulative impact. It also includes

considerations around the roles of designated premises supervisors and personal licence holders.

Section eight considers the breadth of issues affecting public safety. Section nine considers nuisance issues. Section ten considers the issues around the protection of children from harm giving particular thought to preventing access to alcohol and age identification schemes.

This authority is committed to working together with responsible premises management to ensure successful licensed operations. Equally this authority will take appropriate action where there is evidence of irresponsible management. To this end, section eleven of our policy sets out our approach to enforcement, examining how we aim to ensure that our enforcement actions are targeted where they are needed most and co-ordinated together with our partners. It also sets out how we aim to ensure that our enforcement actions are carried out in a fair, consistent and transparent manner.

As this latest revision to our policy is prepared, we are closely monitoring the passage of the Policing and Social Responsibility Bill through parliament, which incorporates many proposed amendments to the 2003 Act. All resultant changes to the licensing regime that arise will be incorporated into this policy in due course.

Aside from this, this policy will be regularly reviewed and will be subject of public consultation at least every three years. We welcome your thoughts on its content and the manner in which we undertake our responsibilities under it.

.

SECTION ONE - ALL ABOUT SOUTHWARK



Our borough

- Alongside the City of London, Southwark is one of the oldest areas of London, with a history stretching back to Roman times. According to statistics produced by the Office of National Statistics (ONT), Southwark's population is estimated to have reached 292,000 in 2009 and is growing. Projections estimate that the population will reach 380,000 by 2031. The population has a young demographic profile and demonstrates rich ethnic and cultural diversity, with around 38% (110,000) of the population from black or ethnic minority communities. Southwark is arguably one of the most diverse areas in the capital.
- 2. Southwark is made up of eight very distinctive urban neighbourhoods that extend along the river Thames and down into South East London. The borough also encompasses some of London's top attractions, creative hotspots, scenic villages and acclaimed green spaces.
- 3. Southwark has a wide-range of leisure and cultural opportunities; and makes a significant economic and employment contribution to the local community. The north of the borough is recognised as one of London's fastest growing tourist quarters and a thriving business location.
- 4. Alongside the borough's rich vibrancy, Southwark has its fair share of challenges. The Index of Multiple Deprivation (IMD) 2007 shows Southwark as the 27th most deprived local authority nationally and 60% of the borough's wards are among the 10% most deprived in the country. Consequently, the borough faces many challenges associated with meeting the complex health and social needs of an inner-city population. Unemployment in Southwark (8.9%) is higher than the London average (6.7%) and the percentage of the working population claiming benefits in Southwark is 15.6% compared to 13.9% across London. Gross weekly earning for both men and women in Southwark is lower than the London average.
- 5. In terms of violent crime, Southwark records a significantly higher number of violence against the person and robbery incidents compared to the London average.
- 6. Whilst there have been improvements, the attainment rates for Southwark pupils at key stages 1 and 2, GCSE and A levels are below the national average. Teenage conception rates for Southwark are still one of the highest in England.
- 7. To meet our challenges, Southwark has a large number of physical regeneration programmes across the borough, alongside a wide range of initiatives aimed at improving educational standards, reducing crime and improving health, housing, social care and the environment.

Leading Southwark

- 8. Southwark is made up of 21 wards, with 63 councillors; three from each ward. These councillors sit on the council assembly, which is the "sovereign" body of the council. The assembly is chaired by the Mayor of Southwark and is responsible for approving the budget and policy framework. Currently labour holds 33 seats; the liberal democrats 25; the conservatives 3 seats; and independent 1 seat. At the time of writing there is one vacant seat.
- 9. The council operates a strong leader / cabinet model. The cabinet is headed by the leader and deputy leader, with eight members holding portfolios of

- Housing management;
- Regeneration and corporate strategy;
- Health and adult social care;
- Transport, environment and recycling;
- Finance, resources and community safety;
- Children's services:
- Equalities and community engagement; and
- Culture, leisure, sport and the Olympics
- 10. The cabinet is responsible for leading the community planning and consultation process, drafting the budget and policy framework, and taking most decisions on resources and priorities.
- 11. Scrutiny committees take a close look at council decisions and activities and call for the council to be open and accountable. They examine services provided by the council and review the council's budget and policies. Scrutiny can look at, or 'call-in' decisions made by the cabinet and recommend decisions are changed or reconsidered.
- 12. A licensing committee is established to consider licensing policy with contested licence applications being considered by sub-committees of the main committee as set down in table 1 in section 4 of this document. Licensing matters fall under the responsibility of the cabinet member for finance, resources and community safety.
- 13. Eight ward based community councils were established in April 2003 to take local decisions in key areas that have a direct impact on the local neighbourhoods.

SECTION THREE – PURPOSE AND SCOPE OF THE LICENSING POLICY

Purpose of the policy

- 14. This policy has four main objectives:
 - To reinforce to elected members on the licensing committee, the powers
 of and constraints placed upon the local authority and to provide the
 parameters under which these licence decisions are made;
 - To inform licence applicants of the parameters under which this authority will make licence decisions and how a licensed premises is likely to be able to operate within an area;
 - To inform residents and businesses of the parameters under which licence decisions will be made and how their needs will be addressed; and
 - To minimise the number of licensing decisions that may be challenged in a court of law.

Scope of the policy

- 15. The Licensing Act 2003 regulates the following activities:
 - The sale by retail of alcohol;
 - The supply of alcohol by or on behalf of a club to, or to the order of, a member of the club;
 - The provision of "regulated entertainment"; and
 - The provision of "late night refreshment"

Types of authorisation

- 16. The Act provides for four different types of authorisation as follows::
 - Personal licences to sell or supply alcohol and / or authorise the sale of alcohol;
 - Premises licences to use a premises for one or more licensable activities; and
 - Club premises certificates to allow a qualifying club to engage in qualifying club activities; and
 - Temporary event notice (TENs) to carry out licensable activities at a temporary event.
- 17. The scope of the policy covers new applications, renewals, transfers and variations of licences including time-limited premises licences. It also covers the review and possible revocation of licences and certificates.
- 18. In general a reference in this policy to a premises licence will also include a club premises certificate.
- 19. This policy does not set out to explain the circumstances under which any one of the available licences or notifications may be required, nor does it attempt to set out the licensing process. This information is provided in separate guidance, available from the licensing service. Please see section 4 of this policy for details.

Definitions

- 20. Under the Act "regulated entertainments" includes entertainment and entertainment facilities. Entertainments include:
 - The performance of a play:
 - The exhibition of a film;

- An indoor sporting event;
- Boxing or wrestling entertainment;
- Performance of live music;
- Any playing of recorded music; and
- A performance of dance;

but only where the entertainment takes place in the presence of an audience and is provided at least partly to entertain that audience

- 21. Entertainment facilities include facilities for:
 - Making music; and
 - Dancing
- 22. "Late night refreshments" constitutes the supply of hot food or hot drink to members of the public from or in the premises, including a vehicle or moveable stall, between the hours of 11pm and 5am for consumption on or off the premises.

The licensing objectives

- 23. In carrying out its licensing functions the council will promote the four licensing objectives set out in the Licensing Act 2003 (2003 Act). They are:
 - The prevention of crime and disorder;
 - Public safety;
 - The prevention of public nuisance; and
 - The protection of children from harm.
- 24. Each licence application will be considered on its own merits in the context of the four licensing objectives. It is important to note that there are no other licensing objectives, so that these four objectives are paramount at all times. Each objective is of equal importance. It is the responsibility of all parties involved in the licensing process to work together toward these licensing objectives at all times. Each of the objectives is considered independently and in more detail in sections 7 to 10 of this document.
- 25. However, the legislation supports a number of other key aims and purposes, which include:
 - The necessary protection of local residents, whose lives can be blighted by disturbance and anti-social behaviour associated with the behaviour of some people visiting licensed entertainment premises;
 - The introduction of better and more proportionate regulation to give business greater freedom and flexibility to meet customers' expectations;
 - Greater choice for consumers, including tourists, about where, when and how they spend their leisure time;
 - The encouragement of more family friendly premises where younger children can be free to go with the family;
 - The further development within communities of our rich culture of live music, dancing and theatre; and
 - The regeneration of areas that need the increased investment and employment opportunities that a thriving and safe night-time economy can bring.

26. These are vitally important and should be the principle aims for everyone involved in the licensing process.

General information

- 27. Licensing is about the control of licensed premises, qualifying clubs and temporary events within the terms of the 2003 Act. Conditions attached to the various permissions will be focused on matters that are within the control of individual licensees and others granted relevant permissions. Accordingly, these matters will centre on the premises and places being used for licensable activities and the vicinity of those premises and places. This will include the direct impacts of the activities taking place at the licensed premises on members of the public living, working or engaged in normal activity in the area concerned.
- 28. However, the licensing function is only one means of promoting the delivery of the objectives described. While we anticipate that the licensing function will make a substantial contribution in relation to licensed premises, it cannot alone resolve all community problems. For instance, the council recognises that licensing law is not the primary mechanism for the general control of nuisance and anti-social behaviour caused by individuals once they are away from licensed premises and beyond the direct control of individual licensees or certificate holders. Nevertheless, it is a key aspect of such control and licensing law will always be a part of the holistic approach to the management of the evening and night-time economy in town and city centres.
- 29. The council will use its full range of powers, engage all relevant responsibilities and work with all partners to deliver the licensing objectives.
- 30. Other mechanisms available for addressing issues of unruly behaviour that can occur away from licensed premises include:
 - Planning controls;
 - The co-ordinating and tactical resourcing roles played by the likes of the partnership tasking group and the public services operation group with the support of the partnership analyst;
 - Ongoing measures such as the council's community wardens; the safer neighbourhood teams; and street leader schemes, all of which aim to create a safe, clean and green environment through partnerships between the council and the community;
 - The provision of CCTV surveillance across the borough; improved street lighting; late night public conveniences; street cleaning and litter patrols;
 - The introduction of the alcohol control area in November 2006, ranging across the borough with the exception of areas of Dulwich, which brings greater powers to penalize and confiscate alcohol from nuisance drinkers;
 - The introduction of the multi-agency partnership night-time economy team;
 - Enforcement powers to deal with statutory nuisance arising from noise, odours and artificial light;
 - Enforcement powers to curb disorder and anti-social behaviour including the issue of fixed penalty notices;
 - Enforcement powers to prosecute personal licence holders or members of staff who sell alcohol to persons who are drunk or sell alcohol to underage children;
 - Enforcement powers to close down instantly, for up to 24 hours, any licensed premises or temporary event on grounds of disorder, the likelihood of disorder or nuisance emanating from the premises causing a

- nuisance. This together with the ability to extend the closure via the magistrates' court and initiate a review hearing where appropriate;
- Campaign work undertaken in conjunction with other partners (such as the Drugs and Alcohol Action Team (DAAT)) under the Safer Southwark Partnership such as previous "safe world cup", "safety glasses"; "talking signs" campaigns;
- 'Conflict resolution training' provided by the council's health and safety team intended for retail staff, aimed at dealing with violence and aggression at work;
- The 'Revealed' training programme run by the council aimed at helping community event organisers provide safe, well managed events;
- The development of partnership arrangements intended to take forward schemes that are of benefit to the local community such as the Business Improvement District in Bankside;
- The establishment of local pub-watch, club-watch and similar schemes.
 Information about pub-watch can be obtained through their web site on www.nationalpubwatch.org.uk; and
- The best practice demonstrated and supported by the Purple Flag, Best Bar None and other similar award schemes.
- 31. While this statement sets out the council's general approach to the making of licensing decisions, nothing in this statement undermines the right of any individual to apply under the terms of the Act for any of the variety of available permissions and to have the application considered upon its individual merits.
- 32. Similarly, this statement of policy does not override the right of any person to make representations concerning the grant or variation of any premises licence or club premises certificate or to seek a review of a licence or certificate where provision has been made for them to do so under the Act. Absolute weight will be given to all representations. However, this will not include those that are frivolous, vexatious or repetitious.
- 33. The council may depart from this policy, if the individual circumstances of any case merit such a decision, in the interests of promoting the licensing objectives. Where such decision is taken, full reasons for the departure from the policy will be given.

SECTION 4 – ADMINISTRATION, EXERCISE AND DELEGATION OF FUNCTION

- 34. As stated in Section 3, this document does not set out to deal with the circumstances under which any one of the available licences or consents may be required, nor does it attempt to explain the applications process.
- 35. Instead, a range of separate guidance documents is available from the Southwark licensing service and can be downloaded from http://www.southwark.gov.uk/info/200127/apply for a licence/519/licensing ac t 2003. These include;

- "Guidance to personal licences";
- "Guidance to premises licences";
- "Guidance to club premises certificates";
- "Guidance to temporary event notices";
- "Guidance to representations";
- "Guidance to licence reviews";
- "Guidance to fees and charges";
- "Guidance to designated premises supervisors" and
- "Public hearings procedure"
- 36. Contact details for the licensing service are provided in section 12 of this document.

Applications for personal licences

37. Applications for the grant and renewal of personal licences must be made on the form prescribed under regulations made by the secretary of state. Applications shall not be progressed until the form has been completed in full and received, together with the relevant fee and all other required information, by the licensing authority. Applications will be progressed in accordance with procedures laid down by the Act. Prospective applicants should refer to the separate document "Guidance to Personal Licences" for further details.

Applications for premises licences

38. Applications for the grant, variation or transfer of a premises licence and applications for a provisional licence, or interim authority, or to change the designated premises supervisor named under a premises licence must be made on the form prescribed under secondary regulations made by the secretary of state. Applications shall not be progressed until the form has been completed in full and received, together with the relevant fee and all other required information, by the licensing authority and other relevant responsible authorities. Applications will be progressed in accordance with the procedures laid down by the Act. Prospective applicants should refer to the separate document "Guidance to Premises Licences" for further details.

Applications for club premises certificates

- 39. In order for qualifying clubs to supply alcohol and / or provide regulated entertainment on club premises, a club premises certificate is required.
- 40. Club premises certificates may be issued to qualifying clubs formerly registered members clubs. To be eligible for a club premises certificate the members must have joined together for a particular social, sporting or political purpose and to purchase alcohol in bulk, for supply to members and their guests. Qualifying conditions are specified in Section 61 of the Act and the council must be satisfied that these conditions have been met. This includes evidence that supports that the club is non-profit making and that there is a minimum of a two-day interval between the application and the acceptance of new members. There is no requirement to specify a designated premises supervisor.
- 41. Applications for the grant or variation of a club premises certificate must be made on the form prescribed under secondary regulations made by the secretary of state. Applications shall not be progressed until the form has been completed in full and received together with the relevant fee and all other required information, by the licensing authority and other relevant responsible authorities. Applications will be progressed in accordance with the procedures

laid down by the Act. Prospective applicants should refer to the separate document "Guidance to Club Premises Certificates" for further detail.

Provision of scale plans

42. Applicants should note that this council requires that current scale plans be submitted with all new applications for premises licences and for any application for a variation of a premises licence affecting approved layout arrangements. Plans must be submitted in scale 1:100 unless otherwise agreed by this authority in writing. Plans need not be professionally drawn but must include all required detail. The information to be included in the plans is set out in secondary regulations and dealt with in the separate guidance produced by this authority. Applicants are asked to note that plans should detail not only the arrangement of internal areas of the premises but also external areas intended to be used as part of the licence. This includes patio or garden areas and any temporary structures such as marquees or smoking shelters.

Variations & new applications

43. Where a premises licence holder wishes to amend their licence the Act allows, in most cases, for an application to vary to be made rather than requiring an application for a new premises licence. A minor variations process exists for proposed variations to a licence that do not impact upon the licensing objectives. A separate major variations process supplements this for variations that carry potential impacts. It should be noted, however, that some proposals to extend the current physical licensed area of the premises concerned may require a new premises licence application to be made. Applicants should discuss their proposals with the licensing team before submitting an application in order to ensure that the correct route is taken.

Provisional statements

44. While applications for a full premises licence may be made in respect of a premises that is not yet complete, this council would prefer to see provisional statements sought in the first instance in cases where the responsible bodies are not yet able to confirm that works have been carried out in compliance with agreed schedules and to standard. However, the council acknowledges that any person falling within section 16 of the Act can apply for a premises licence before new premises are constructed, extended or changed, as long as clear plans of the proposed structure exist and the applicant is in a position to complete an operating schedule. The schedule will need to contain details of the activities to take place there; the time at which such activities will take place; the proposed hours of opening; where the applicant wishes the licence to have effect for a limited period, that period; the steps to be taken to promote the licensing objectives; and where the sale of alcohol is involved, whether supplies are proposed to be for consumption on or off the premises (or both) plus the name of the designated premises supervisor.

Temporary event notices (TENs)

45. The Act provides a system by way of "temporary event notices" for the temporary carrying on of any licensable activity outside of the terms of a premises licence or club premises certificate. Under this system no authorisation as such is required from the licensing authority. Instead the premises user notifies the event to the licensing authority and the police. In general, only the police may object (on crime and disorder issues) to such an event from taking place. The licensing authority may only intervene of its own volition, if the notice is incorrect or if the limits set out in the Act on the number of events that may be given are exceeded.

- 46. It is important to recognise that TENs may be given for the occasional extension of licensed hours or activities at currently licensed premises, as well as for events at unlicensed premises.
- 47. TENs are subject to various conditions and limitations:
 - The duration is limited to a period of up to 96 hours;
 - Events cannot involve the presence of more than 499 people at any one time:
 - The same premises can only be used for up to 12 occasions in a calendar year, but the aggregate number of days must not exceed 15;
 - There must be a minimum of 24 hours between event periods for the same premises by the same premises user;
 - A personal licence holder is limited to 50 events per year; and
 - A person not holding a personal licence is limited to 5 events per year.
- 48. If the above conditions are not fulfilled then a temporary event at which licensable activities are to take place would require a premises licence.
- 49. Temporary event notices must be given on the form prescribed under secondary regulations made by the secretary of state. Notifications shall not be progressed until the completed form has been received by the licensing authority and the police, and the relevant fee paid.
- 50. The law states that at least 10 working days notice must be given to the licensing authority and the police. No relaxation of this requirement may be given. "Ten working days notice" means ten working days exclusive of the day on which the event is to start, and exclusive of the day on which the notice is given. So, if a temporary event is intended for a Saturday or Sunday night, the notice must be given three Fridays before the event date at the latest. If a bank holiday should fall within the period then an additional working day must be allowed for each bank holiday day that occurs.
- 51. As stated, the police may object to the event, within two working days from when they are given the notice, for reasons of preventing crime and disorder. In a significant number of events it is anticipated that the minimum notice period will not provide sufficient time for proper consideration to be given to any crime and disorder concerns that the police may raise. For this reason, while this authority will accept notifications given with only the statutory minimum notice, it is requested that 28 days notice of an intention to stage a temporary event is provided, together with as much supporting information as possible, in order to prevent unnecessary objections having to be raised by the police.
- 52. It should also be noted that in cases where there is any element of doubt as to whether the premises user has the consent of the premises owner to use the premises concerned for the purposes given; this authority will wish for both the police and itself to be provided with evidence to this effect.
- 53. Furthermore, it should be recognised that many temporary events will still have potential to give rise to concerns, particularly around noise nuisance. For this reason applicants should understand that the licensing service may share information on temporary events taking place with other "responsible authorities". This does not provide those responsible authorities with the opportunity to raise representations but will allow any matter of concern to be followed up under other powers. This action does not, however, absolve any premises user from their own responsibility to ensure that any other necessary

consent (including planning permission) is obtained or that their event does not give rise to any safety, nuisance or harm to children issues. Further guidance is available in the separate document "Guidance to Temporary Event Notices" available from the licensing service.

- 54. Aside from the TENs system, it should be noted that in the event that the environmental protection team considers that a nuisance is likely to be caused by any temporary event, a noise abatement notice may be served on the appropriate person(s) under the provisions of the Environmental Protection Act 1990 to prevent such nuisance occurring.
- 55. Organisations looking to host a temporary event can contact the environmental protection team for advice on how to best reduce the risk of causing a nuisance.

Major art and pop festivals, carnivals, fairs and circuses

56. We encourage organizers of major festivals and carnivals to approach the council at the earliest opportunity to discuss arrangements for the licensing of their events. On occasions event organizers may wish to seek a single time-limited premises licence to cover a wide range of activities across a broad footprint. This will involve the preparation of a substantial operating schedule. The Southwark events team co-ordinates a safety advisory group with membership comprising members of all the emergency response authorities which can provide support to event organisers on event planning if good time is allowed. Contact details for the events team are provided in section 12

Consultation arrangements

- 57. Southwark council considers it important that the local community is fully aware of local licensing applications being made within its area. Regulations governing the advertising of applications for the grant, variation or review of premises licences and club premises certificates require a public notice to be published in the local press and a brief summary of the application to be displayed immediately on or outside premises concerned. Failure to comply with the requirements in full will result in the progression of the application being delayed.
- 58. This authority understands that regulations to the Act require public notices to be of a size equal or larger than A4; of pale blue colour; and printed legibly in black ink or typed in black in a font of a size equal to or larger than 16. It also recognises that applications may not be rejected on the basis of improper notification if public notices comply with this minimum standard. However, in order to ensure that the notice of application displayed at the premises may be clearly seen and easily read by passers by, this authority recommends that public notices be of A3 size in pale blue colour printed legibly in black ink in a font of size equal to or larger than 20. This authority also asks that the notice summarises the proposed licensable activities and the proposed hours of opening and remains clearly displayed in a position where it may be readily seen by passers by for the period during which representations may be made. Checks will be made to ensure that the summary is properly displayed.
- 59. The licensing service supports the consultation process by maintaining a public register at http://app.southwark.gov.uk/licensing/licenseregister which details all current licence applications under consideration (alongside detail of currently licensed premises). It is open to any person who wishes to be kept abreast of latest applications in their local area, to subscribe to "licensing alerts". E-mail notifications are sent to those who have registered when a new licence

application is made within the relevant ward. Furthermore, the licensing service will ensure that local ward councillors are informed of all new applications in their area.

Responsible authorities

60. Contact details for responsible authorities are given in section 12 of this policy. They include public bodies that must be notified of, and that are entitled to lodge representations about, applications for premises licences or club premises certificates or major variations. They may also ask a licensing authority to review a premises licence or club premises certificate. Responsible authorities will also provide advice and information on established best management practice. Some information on preparing an operating schedule is contained within this policy but applicants may also wish to consult with the authorities directly.

Interested parties

- 61. Interested parties are defined under the Act:
 - A person who lives or a body which represents persons who live in the vicinity of the premises in question;
 - A person involved in a business in that vicinity or a body which represents persons involved in such businesses; and
 - A member of the relevant licensing authority.
- 62. This authority has not attempted to define 'vicinity'. Any issues of vicinity will be considered on a case by case basis upon it's merits.
- 63. The definition of "interested parties" includes local ward councillors. While ward councillors may make representations by right they may also make representations in writing and at a hearing on behalf of another interested party such as a resident or local business where specifically authorised to do so.
- 64. In recognition that some interested parties may feel disadvantaged in the representations process, this council wishes to make the process of authorizing local ward councillors and other representatives to speak on behalf of other interested parties as easy as possible. For this purpose pro-forma authorization forms may be downloaded from www.southwark.gov.uk/businesscentre/licensing or obtained from the licensing service. If using the authorization process you should firstly obtain the consent of the person you intend to represent your views and then provide a completed and signed authorization form to the licensing service before the last date for submission of representations.
- 65. It should be understood that applicants will normally retain the right to know the name and address of any person who lodges a representation, even if an authorisation form has been completed. However, this authority recognises that in some exceptional and isolated circumstances interested parties may be reluctant to make representations because of fears of intimidation or violence if their personal details are divulged. Where it is considered that the interested party has a genuine and well-founded fear of intimidation and may be deterred from making a representation because of this, this authority will consider withholding some or all of the interested parties' details.

Representations

- 66. It is open to any "responsible authority" or "interested party", as defined under the Act, to lodge representations concerning a premises licence club premises certificate application during the set consultation period.
- 67. A representation will only be "relevant" if it relates to the likely effect of the grant of the licence on the promotion of at least one of the licensing objectives. A representation that fails to do this is not "relevant" for the purposes of the 2003 Act. It is for the licensing authority to determine on its merits whether any representation by an interested party is frivolous or vexatious.
- 68. Representations must be received in writing at the licensing service office by the last date for representations. Representations must be signed, dated, provide the grounds of representation in full and include the name and address of the person / body making the representation.

Determination of applications

- 69. In accordance with the provisions of the licensing act this authority has established a licensing committee with a maximum of 15 members and provision for sub-committees of three members to be drawn from the overall pool of the main committee membership.
- 70. In doing so, the council has sought to establish a speedy, efficient and cost-effective system for determinations of licences. Where the Act carries a presumption of grant for applications to which no objection has been made, the administration of the grant will be delegated to officers. All such matters will be periodically reported for information only to the full licensing committee.
- 71. Where matters are subject of representations, the licensing service will normally attempt to reach a negotiated conclusion through a conciliation process. This process is operated in recognition of the fact that the most workable solutions are normally ones that all parties concerned can sign up to and in order to attempt to save the time and costs associated with holding a public hearing. Conciliation can be agreed up until 24 hours before a hearing takes place. Where this is not possible the application and representations to it will be considered by the sub-committee at a public hearing.
- 72. Table 1 sets out the schedule of delegation of decisions and functions to licensing committee, sub-committees and officers. This form of delegation is without prejudice to the council's right to redirect an application as appropriate in the circumstances of any particular case.

Table 1 – Table of delegation	s of licensing fur	nctions	
Matter to be determined	Full committee	Sub-committee	Officers
Licensing policy issues	& assembly		
Application for personal		If a police	If no
licence		representation	representation
Application for personal		If a police	If no
licence with unspent		representation	representation
convictions			
Application for premises		If a relevant	If no relevant
licence / club certificate		representation	representation
		made	
Application for a provisional		If a relevant	If no relevant
statement		representation	representation

	made	
Application for a 'major'	If a relevant	If no relevant
variation of a premises	representation	representation
licence / club certificate	made	
Application for a 'minor'		All cases
variation of a premises		
licence / club premises		
certificate		
Application to vary the	If a relevant	If no relevant
designated premises	representation	representation
supervisor	made	
Request to be removed as		All cases
designated premises		
supervisor	If a relevant	If no relevant
Application for transfer of premises licence		
premises licence	representation made	representation
Application for interim	If a relevant	If no relevant
authority	representation	representation
-	made	
Application to review	All cases unless	
premises licence / club	application and	
certificate	representations	
5	are withdrawn	A.II
Decision on whether a		All cases
complaint is irrelevant,		
frivolous, vexatious etc		All cases
Decision to object when a local authority is a consultee		All Cases
and not the lead authority		
Determination of police	All cases	
representations to a	All Cases	
temporary event notice		
tomporary event notice		

Licence reviews

- 73. At any stage following the grant of a premises licence, a responsible authority or an interested party, may ask the licensing authority to review the licence because of a matter arising at the premises in connection with any of the four licensing objectives. The licensing authority itself cannot initiate the review process.
- 74. In addition, a review of the licence will normally follow any action by the police to close down a premises for up to 24 hours on grounds of disorder or nuisance, as a result of a magistrate's' court's determination sent to the licensing authority. A licence review can have several outcomes, ranging from no action being found necessary to the temporary suspension or revocation of a licence.
- 75. Where a review of a premises licence is sought this must relate specifically to a particular premises licence relating to an individual premises. Guidance given above on frivolous and vexatious applications applies in respect of applications for reviews of licences also. The licensing authority may also reject an application made by an interested party if it determines that the ground for the review is repetitious, that it is substantially similar to an earlier review, following which a "reasonable interval" has not elapsed since the earlier application or alternatively since the grant of the premises licence.

76. Further information, including an outline of the process that is followed at licence review hearings, can be found in the separate document "Guidance to Licence Reviews".

2012 Olympics

77. The council is fully committed to a safe and successful Olympic and Paralympics games in London in 2012. The council recognises the resources of the police and the emergency services will be planned out and prioritised for the security of major events before, during and after the games from May 1 2012 until October 31 2012. Due consideration will be given by the council to representations from the police or community safety officers, in relation to licence applications for activity during the games time on the grounds of public safety and security when police and other emergency services resources are insufficient to deal with the risks presented. Where, as a result of representations from a responsible authority, it is identified that a licence or proposed event presents a risk that the licensing objectives will be compromised, it is the council's policy that such applications will not be granted.

SECTION 5 – DETERMINING APPLICATIONS FOR PREMISES LICENCES AND CLUB PREMISES CERTIFICATES

How this policy works

- 78. All applications for new premises licences or variations need to be supported by an operating schedule. The schedule must specify (among other things) the steps which the applicant proposes to promote the licensing objectives.
- 79. If an application for a premises licence or club premises certificate has been made lawfully and there have been no representations from responsible authorities or interested parties, the licensing authority must grant the application, subject only to conditions that are consistent with the operating schedule and relevant mandatory conditions. The licensing authority will have no discretion to refuse the application or to alter or add to the conditions arising from the operating schedule.
- 80. Where, however, there are relevant representations, then (unless the concerns raised within the representations are conciliated) a hearing of the opposed application before the licensing sub-committee will normally follow. After the hearing, the sub-committee must, having regard to the representations, take such steps as it considers necessary to promote the licensing objectives. These

- may include refusing the application, or adding to or modifying the conditions proposed in the operating schedule.
- 81. In exercising its discretion, the licensing sub-committee will have regard (amongst other things) to this licensing policy. Therefore, in drawing up their operating schedule, applicants would be well advised to read this policy carefully. Where an operating schedule complies with this policy, it is generally less likely that an interested party or responsible authority will object to it, or that any representation will succeed. Therefore, compliance with this policy is likely to assist the applicant to avoid the delay and expense of a contested licensing hearing, and the risk of a refusal or the addition of unwanted licence conditions.
- 82. This is not to say that an opposed application which complies with the policy will necessarily be granted or that an opposed application which does not comply with it will necessarily be refused. Where there have been relevant representations, the licensing authority will always consider the merits of the case, and interfere with the operating schedule only when, and to the extent, necessary to promote the licensing objectives. Nor will blanket or standard conditions be applied without regard to the merits of the individual case. So, for example, following receipt of relevant representations the licensing authority will not interfere with an operating schedule which does not comply with this policy where the steps proposed are sufficient to meet the licensing objectives in the individual circumstances of the case.
- 83. However, the policy represents the licensing authority's view of the best means of securing the licensing objectives in most normal cases. It has been drawn up in consultation with other expert bodies and responsible authorities, together with community stakeholders. While the contents of the operating schedule are a matter for the applicant, where there is objection to a schedule which departs from the policy, the licensing sub-committee hearing an opposed application will normally expect to be given a good reason for the departure if it is to be asked to make an exception to the policy.
- 84. In this policy, there are a number of references to the licensing authority's expectations of applicants. As explained above, the policy is only engaged where the licensing authority has a discretion following the receipt of objections. In such cases, the licensing authority will not apply the policy rigidly, but will always have regard to the merits of the case with a view to promoting the licensing objectives.
- 85. One particular expectation of applicants is that they conduct a risk assessment in relation to the licensing objectives before completing an operating schedule. It is emphasized that there is no statutory requirement for this under the Licensing Act 2003. However, where no such risk assessment has been conducted then if there are relevant representations leading to a hearing, the licensing sub-committee may take a more precautionary approach than if the applicant can demonstrate that a particular risk has been properly evaluated and either discounted or mitigated in the operating schedule.

Conditions

86. Key concepts underscoring the Act are that each application must be considered upon its own merits and any conditions attached to licences and certificates must be tailored to the individual style and characteristics of the premises and events concerned. This is considered essential to avoid the imposition of disproportionate and overly burdensome conditions on premises

- where there is no need for such conditions. To this end, following relevant representations, the Council will not impose blanket standard conditions.
- 87. As stated above, where an operating schedule or club operating schedule has been submitted with an application and there have been no relevant representations made by responsible authorities or interested parties, the licence or certificate must be granted. In such cases, the licence will be made subject only to conditions as are consistent with the schedule accompanying the application and any mandatory conditions required by the Act itself.
- 88. Where relevant representations have been made and following a hearing, the licensing authority may determine that the steps needed to promote the licensing objectives proposed by the applicant should be modified. In such cases, the licensing authority may revise the conditions expressed in the operating schedule or club-operating schedule in terms that it considers necessary.
- 89. Where this discretion is engaged, the council will consider attaching appropriate and proportionate conditions. Reference is made at various points in this document to potential management controls and conditions. Where such conditions are to be applied, it will be on the basis that they are appropriate to suit the specific needs of an individual premises operation. Alternative conditions may also be specifically tailored by the council and attached to licences as appropriate.
- 90. In order to minimise the need for representations to be lodged by responsible authorities because of insufficient regard being had to the preparation of operating schedules it is recommended that applicants consult with responsible authorities when operating schedules are prepared.

Duplication

91. So far as possible, this policy is not intended to duplicate other existing legislation and regulatory regimes. To this end, conditions shall only be attached to premises licences that are not are provided for within other legislation and are necessary for the promotion of the licensing objectives.

SECTION 6 - OTHER RELATED LEGISLATION AND STRATEGIES

92. This policy recognises and intends to complement the broad range of other relevant legislation and council and government policy, strategy, responsibility, and guidance.

Other relevant legislative requirements

93. The council has other general statutory requirements relating to crime and disorder, anti-social behaviour, health, nuisance, human rights, disability discrimination and race relations, all of which complement the Licensing Act 2003. These are briefly outlined below.

Crime and Disorder Act 1998

- 94. Section 17 of the Crime and Disorder Act 1998 is aimed at giving the vital work of crime and disorder reduction a focus across the wide range of local services and putting it at the heart of the local decision-making. It places a duty on key authorities, including local authorities and the police and fire and rescue authorities, to do all they reasonably can to prevent crime and disorder in their area.
- 95. All local authorities must fulfil their obligation under section 17 of the Act when carrying out their functions as licensing authority under the 2003 Act. The council as licensing authority must consider how the exercise of its functions impacts upon and contributes toward the prevention of crime and disorder within its area.

96. This statement of licensing principles is intended to support the work of the local Crime and Disorder Reduction Partnership, the Safer Southwark Partnership through measures under both the **Crime and Disorder Reduction Strategy** and the **Violent Crime Reduction Strategy**.

The Anti-Social Behaviour Act 2003

97. The Anti-Social Behaviour Act 2003 identifies types of anti-social behaviour that affect community life and provides local authorities with the tools to deal with these. Of particular note, sections 40 and 41 of the Act provide local environmental health officers complementary powers to those of the police under part 8 of the Act, to close licensed premises causing a public nuisance for temporary periods

Violent Crime Reduction Act 2006

98. The Violent Crime Reduction Act 2006 received Royal Assent on 8 November 2006. The Act introduced new measures to ensure that police and local communities have the powers they need to tackle guns, knives and alcohol related violence. The Act also provided further powers to local licensing authorities in relation to crime and disorder.

Policing and Crime Act 2009

- 99. The Policing and Crime Act 2009 clarifies how police forces and local authorities can work together by placing an explicit duty on police authorities to reflect their priorities in their work. The Act introduced a number of measures which are relevant or made changes to the 2003 Act, including:
 - A mandatory code of practice for alcohol retailers;
 - Elected members of licensing authorities as interested parties; and
 - Reclassification of lap dancing clubs so they require a sex establishments licence.

The Health Act 2006

100. The Health Act 2006 introduced the ban on smoking in all enclosed workplaces and public spaces, including smoking in pubs, restaurants and members' clubs where bar or other staff are employed. The law is enforced by local authorities.

The Clean Neighbourhoods and Environment Act 2005

101. This provides local authorities with an additional power to issue fixed penalty notices to any licensed premises emitting noise that exceeds the permitted level between the hours of 11pm and 7am.

The EU Services Directive

- 102. The EU Services Directive is a flagship European Directive intended to develop the single market for services by breaking down barriers to cross-border trade within the EU and making it easier for service providers within scope to set up business or offer their services in other EU countries. The Directive requires that all notices and authorisations in scope are able to be completed electronically and via a 'single point of contact'. The Directive was implemented by the Provision of Services Regulations in December 2008. The UK point of single contact is the Electronic Application Facility (EAF) which is part of the www.businesslink.gov website.
- 103. Although only regulated entertainment is a 'service' as defined under the Directive, the Government has extended the electronic application process to all regulated activities under the 2003 Act and to all authorisations and notices

with the exception of applications for, and renewals of personal licences, reviews and representations.

The Human Rights Act 1998

- 104. The Human Rights Act 1998 incorporates the European Convention on Human Rights, making it unlawful for a local authority to act in any way that is incompatible with a convention right. The council will have particular regard to the relevant provisions of the European Convention of Human Rights
 - Article 1 of the first protocol that every person is entitled to the peaceful enjoyment of his possessions, including for example the possession of a licence;
 - Article 6 that in determination of civil rights and obligations everyone is entitled to a fair and public hearing within a reasonable time by an independent and impartial tribunal established by law; and
 - Article 8 that everyone has the right to respect for his home and private life.

Equality Act 2010

- 105. From 1 October 2010, the Equality Act 2010 (EA) replaced most of the Disability Discrimination Act (DDA). The Equality Act 2010 aims to protect disabled people and prevent disability discrimination. It provides legal rights for disabled people in the areas of:
 - Employment;
 - Education;
 - Access to goods, services and facilities including larger private clubs and land based transport services;
 - Buying and renting land or property; and
 - Functions of public bodies, for example the issuing of licences.
- 106. The Equality Act also provides rights for people not to be directly discriminated against or harassed because they have an association with a disabled person. This can apply to a carer or parent of a disabled person. In addition, people must not be directly discriminated against or harassed because they are wrongly perceived to be disabled.
- 107. This council firstly recognises that it has a duty to ensure that no disabled person is
 - Treated less favourably because they are disabled; nor
 - Indirectly discriminated against unless there is clear reason to do so.
- 108. As will be noted from the contact details provided in section 12 of this document, contact can be made with our licensing and related services in a number of ways. Our customer service centre is available 24/7. Our services can receive telephone; email and postal communications. Advice and information is provided on our web site or direct from our offices. Officers will make personal calls by appointment. If you require our services and have difficulties using our services please talk to us and explain exactly what your needs are to help us understand what adjustments may be necessary to the way we provide our services.
- 109. We also recognise that other service providers providing licensable activities including shops, hotels, pubs, theatres and more hold similar responsibilities.

Service providers have an obligation to make reasonable adjustments to premises or to the way they provide a service. Sometimes it just takes minor changes to make a service accessible.

- 110. While access for people with disabilities is not one of the four licensing objectives, this council expects that responsible licensees will comply with the requirements of the Equality Act 2010. As such the licensing service will work closely with the Southwark Disablement Association (SDA) to ensure that disabled people are not treated less favourably than other people for a reason related to their disability. As part of this arrangement this authority will inform the SDA of new licence applications and will support the SDA and licensees in improving access to services. The SDA can provide advice to any licensee on reasonable adjustments contact details are provided in section 12 of this policy.
- 111. Alternatively, the Directgov website provides advice on access to services at http://www.direct.gov.uk/en/DisabledPeople/Everydaylifeandaccess/Everydayaccess/DG_4018353. The web site also includes links to DisabledGo and Direct Enquiries, which are online directories with detailed access information about venues across the UK. You can search the database, and filter results so that you can check whether a venue is suitable for your own individual needs.

The Race Relations Act 1976

- 112. The Race Relations Act 1976, as amended by the Race Relations (Amendment) Act 2000, places a legal obligation on public authorities to have due regard to the need to eliminate unlawful discrimination; and to promote equality of opportunity and good relations between persons of different racial groups.
- 113. This council recognises Southwark's diversity as one of it's strengths and is committed to creating a more inclusive community. The council will aim to ensure that nothing within this Statement of Licensing Policy nor it's associated practices discriminates against any group within the community and will pursue any opportunity to promote equality of opportunity and good community relations. To this end the council will undertake equalities impact assessments of the policy at regular intervals to identify the effects the policy may have had for different groups and take appropriate action to prevent direct and indirect discrimination which may have been the result of the policy statement.

Other strategies and policies

- 114. The council will look to secure the proper integration of this policy with other council and government policies, strategies, responsibilities, and guidance documents issued.
- 115. In this context this policy is informed by the following strategies and policies.

Alcohol Harm Reduction Strategy

- 116. Southwark's local Alcohol Harm Reduction Strategy, prepared in response to the **National Alcohol Harm Reduction Strategy** ("Safe. Sensible. Social. The next steps in the National Alcohol Strategy" (June 2007)) explores the relationships between alcohol and crime alongside health impacts.
- 117. In preparing this statement the council is particularly aware that the national strategy has identified a number of priorities that should be considered when taking decisions in connection with licensing matters.

- To reduce the problems arising from under-age drinking;
- To reduce public drunkenness; and
- To prevent alcohol related violence.

Enforcement policy

118. All enforcement actions taken by the licensing service are governed by the council's enforcement policy, which has been drafted with full reference to the enforcement concordat and the provisions of the BERR Statutory Code of Practice for Regulators, the Police and Criminal Evidence Act 1984 (PACE) and the Regulation of Investigatory Powers Act 2000 (RIPA).

Cultural strategy

- 119. As part of implementing local cultural strategy, the council will actively encourage and promote a broad range of entertainment, particularly live music, dancing and theatre, including the performance of a wide range of traditional and historic plays for the benefit of communities.
- 120. While this authority maintains a natural concern to prevent disturbance in neighbourhoods, it also recognises the wider cultural benefits for local communities. In determining what conditions should be attached to licences and certificates for the promotion of the licensing objectives, this authority will be aware of the need to avoid measures which deter live music, dancing and theatre by imposing indirect costs of a disproportionate nature. Live performance is central to the development of cultural diversity and vibrant and exciting communities, where artistic freedom of expression is a fundamental right and greatly valued. Traditional music and dancing are parts of the cultural heritage of England and Wales. Music and dancing also helps to unite communities and particularly in ethnically diverse communities, new and emerging musical and dance forms can assist in the development of a fully integrated society. It should also be noted that the absence of cultural provision in an area can itself lead to the young people being diverted into anti-social activities that damage communities and the young people themselves.
- 121. The council has also pursued a policy of seeking premises licences for its open spaces in its own name. Where such licences are in place, performers and entertainers may not need to obtain a licence or give a TENs themselves in order to perform. Instead they would require permission from the council as premises licence holder. Southwark's Leisure Services has produced the "Community Event Organiser's Toolkit" to support independent activities. Details of licences held by community venues may be viewed at http://licensing.southwarksites.com

Planning & building regulations regimes

- 122. Lastly, it should be made clear that the planning, building control and licensing regimes will be properly separated to avoid duplication and inefficiency.
- 123. It is recognised that licensing applications should not be a re-run of the planning application and should not cut-across decisions taken by the local authority planning committee or permissions granted on appeal. Similarly, the granting by the licensing committee of any variation of a licence which involves a material alteration to a building would not relieve of the need to apply for planning permission or building control where appropriate.
- 124. While the licensing authority is conscious that there is no legal basis for a licensing authority to refuse a licence application solely because it does not have planning permission, it may be inconsistent for the authority to give a

licence for an activity when it has refused planning permission for the same activity to take place. It is anticipated thereby that applications for premises licences for permanent commercial premises should normally be from businesses with planning consent for the property concerned. Where this is not the case the council would expect the applicant to address the reasons why planning permission had not been granted and provide reasons as to why licensing consent should be.

- 125. There may also be circumstances when as a condition of planning permission, a terminal hour has been set for the use of the premises for commercial purposes which carries different hours to the licensing hours. Where these hours are different to the licensing hours, the applicant must observe the earlier closing time. Premises operating in breach of their planning permission would be liable to enforcement under planning law.
- 126. Borough-wide and area based planning guidance exists through the **Southwark Plan** and supplementary planning guidance. Where relevant representations are made on applications, the Licensing Authority will have regard to the appropriate relevant guidance.
- 127. Proper integration with the planning regime will therefore be assured. This will include, where appropriate, the provision of regular reports to the planning committee on the situation regarding licensed premises in the area, including the general impact of alcohol related crime and disorder.
- 128. The council is also aware of the Mayor of London's best practice guidance on "Managing the Night Time Economy" and will have regard to its contents.

Guidance documents

- 129. This policy is also mindful of the direction provided under the following guidance documents:
 - The Home Office Safer Clubbing Guide issued to ensure the health and safety of anyone who attends dance events, with a particular emphasis on those who also take controlled drugs;
 - The LACORS Practical Guide to Test Purchasing which deals with the involvement of children in the enforcement of legislation concerning the sale of age-restricted goods; and
 - Furthermore, the objectives of the Private Security Industry Authority (SIA)
 will be taken into account in all issues relating to the provision of door
 supervision at licensed operations.
- 130. This policy will also be checked for consistency with all current legislation, strategy, policy and guidance on an annual basis.

SECTION 7 – THE FIRST LICENSING OBJECTIVE – THE PREVENTION OF CRIME AND DISORDER

Note: For how this policy is intended to apply see Section 5

- 131. The council recognises that well run licensed premises can make a valuable positive contribution to the local community through the local economy, tourism, and cultural development. However, If they are not properly managed and controlled, licensed premises, especially those offering late night entertainment, or alcohol and refreshment for large numbers of people, can become a serious source of crime and disorder or anti-social behaviour problems.
- 132. Thereby, the Council will expect applicants for premises licences to demonstrate in their operating schedules that suitable and sufficient steps are to be taken to prevent crime and disorder in and around the vicinity of their premises and events.
- 133. Operating schedules accompanying premises licence and club premises certificate applications should be prepared on the basis of a risk-assessment of the potential sources of crime and disorder. Issues of crime and disorder can take many forms. Relevant issues might include:
 - Underage drinking;
 - Drunkenness on the premises;
 - Public drunkenness;
 - Drugs use:
 - Violent behaviour, including the possession of weapons;
 - Anti-social behaviour;
 - Counterfeit goods;
 - Theft of personal property;
 - Prostitution, lewd acts and similar offences;
 - Tax evasion; and
 - Maintenance of smoke-free environments.

- 134. Applicants are recommended to seek advice from the council's licensing officers and police when carrying out their risk assessments and preparing their operating schedules.
- 135. Table 2 is provided, together with a key, across pages 33-35. It is intended as a helpful broad guide to applicants as to the types of controls that may be considered when assessing the steps necessary for the promotion of the prevention of crime and disorder licensing objective, following the receipt of relevant representations. The key to table 2 also provides suggested wording for an appropriate condition.
- 136. It is emphasised that this is neither meant to be a definitive list of possible preventative measures nor will the controls listed be necessary in every circumstance. It is recognised that applicants know their premises and method of operation and controls proposed under the operating schedule should properly reflect the proposed operation of the premises concerned.
- 137. While it is recognised that there is no requirement in the 2003 Act to do so, applicants may find that it helps the licensing process to provide a copy of their completed risk-assessment upon which the operating schedule is based to the police for consideration at the time of application.
- 138. Where its discretion is engaged following consideration of relevant representations, the licensing authority will consider attaching relevant conditions to licences and permissions to deter crime and disorder, both inside and immediately outside the premises. As far as possible these will reflect local crime prevention strategies.
- 139. As an additional aide to good management, this authority recommends the range of qualifications designed to support the licensing objectives, available from the British Institute of Innkeeping (BII). Further information is available by contacting the BII by e-mail at the following address info@bii.org

Mandatory conditions

- 140. A number of mandatory licence conditions are set out in the Act. The following mandatory conditions will apply to all premises licences authorising the retail sale of alcohol (with the exception referred to in paragraph 142 below).
 - (a) Where a premises licence authorises the sale of alcohol, every supply of alcohol is to be made or authorised by a person who holds a personal licence; and
 - (b) Where a premises licence authorises the supply of alcohol, no supply may be made if no designated premises supervisor has been specified in the licence, or at a time when the designated premises supervisor does not hold a personal licence, or his or her licence has been suspended.
- 141. Note: The Legislative Reform Order (Supervision of Alcohol Sales in Church and Village Halls) Order 2009 allows for certain community venues to apply for an alternative licence condition the effect of which will be that the licence holder (i.e. the management committee) becomes responsible for the supervision and authorisation of all sales.
- 142. Where a premises licence includes a condition that at specified times one or more individuals must be at the premises to carry out security activities, the

- licence must contain a condition that such individuals must be licensed by the Security Industry Authority (SIA).
- 143. Furthermore, five new mandatory conditions were introduced under the Licensing Act 2003 (Mandatory Licensing Conditions) Order 2010. These new mandatory conditions cover:
 - The prohibition of irresponsible drinks promotions (on sales only);
 - Restrictions on one person dispensing alcohol directly into the mouth of another (on sales only);
 - The provision of free tap water (on sales only);
 - Requirements for age verification policies to be established in relation to the sale and supply of alcohol, including photo-identification for persons appearing to be under 18 years of age (on and off sales); and
 - Requirements for alcohol to be sold or supplied in small measures (on sales

	Table 2 - Broad guide to the range of crime and disorder controls that might be considered under various categories of premises operation (see key provided by figure 1)																							
	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	20	21	22	23	
Off licences / shops / supermarkets		Y	Υ	Υ	Υ																	Υ	Y	
Community halls and similar facilities / Members & Sports Clubs	Y	Y	Υ		Y	Y	Y									Y						Y	Y	
Restaurants / cafes	Υ	Υ	Υ	Υ	Υ	Υ																Υ	Υ	
Cinemas / theatres	Υ	Υ	Υ	Υ	Υ	Υ	Υ												Y		Y	Y	Υ	
Pubs and bars	Υ	Υ	Υ	Υ	Υ	Υ	Υ	Υ				Υ			Υ	Υ	Υ	Υ	Υ			Υ	Υ	
Entertainment bars	Υ	Υ	Υ	Υ	Υ	Υ	Υ	Υ	Υ	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y	
Night Clubs	Υ	Υ	Υ	Υ	Υ	Υ	Υ	Υ	Υ	Υ	Υ	Υ	Υ	Υ	Υ	Υ	Υ		Υ	Υ	Υ	Υ	Υ	
Vessels	Υ	Υ	Y	Υ	Υ	Υ	Υ	Υ	Υ	Υ		Υ	Υ			Υ	Υ		Υ	Υ	Υ	Υ	Υ	

Key 1 – Broad Guide to the level of Crime and Disorder Controls that may be considered under various categories of premises operation

- Set a safe capacity limit While often necessary on public safety grounds, this should also be considered where overcrowding may lead to risk of violence or disorder. Example condition "That the maximum number of persons that may be accommodated in the (state area of premises concerned) .. shall not exceed ... (state safe maximum number of persons)"
- **Use of crime prevention notices** for use in circumstances where it would be helpful to warn customers of prevalence of crime which may target them, for instance, to warn of pickpockets of bag snatchers. Example condition "That crime prevention notices will be used to provide relevant crime prevention information to customers as deemed appropriate by management in consultation with the police"
- 3 Publicise details of the premises operation "That details of the premises opening and closing times will be clearly displayed upon the premises for the information of customers"
- 4 Install CCTV The presence of CCTV cameras can be an important means of detecting crime at and immediately outside licensed premises. Example conditions "That CCTV equipment shall be installed upon the premises to the satisfaction of the local police licensing office and shall be maintained in good working order" and "That a library of CCTV recordings shall be maintained for 31 days and made available to any authorized officer for inspection"
- **Maintain control over the removal of open containers –** To prevent the use of containers as offensive weapons in surrounding streets after individuals have left the premises. Example condition "That no open containers may be removed from the premises and signs shall be exhibited to this effect"
- **6 Use of plastic containers and toughened glass –** Consideration should be given to the use of safer alternatives to glass which would inflict less serious injuries if containers were used as weapons. "That drinks will only be served in non-breakable plastic containers or toughened glass. Plastic glasses used for draught beer and cider must be legally marked unless a crown stamped beer measuring instrument is in use"
- Introduce bottle bans Example conditions "No bottles containing beverages of any kind, whether open or sealed, shall be given to customers on the premises whether at the bar or by staff away from the bar"; "No customers carrying open or sealed bottles shall be admitted to the premises at any time that the premises are open to the public" (though care should be taken where off-sales are provided or bottles of wine for consumption with a table meal)
- **Employ SIA registered door supervisors** Valuable for maintaining orderly behaviour in queues; searching and excluding those suspected of carrying offensive weapons or illegal drugs; keeping out banned individuals; or controlling admissions and departures. Example condition "That (an adequate number) of SIA registered door supervisors will be employed at the entrance of the premises to monitor admissions to and departures from the premises and at other key internal management control points as necessary for the purposes of security, protection, screening and dealing with conflict"
- 9 Make personal searches by door supervisors a condition of entry "That all admissions and re-admissions to the premises, including by staff and performers, will be subject to a personal search by SIA registered door supervisors. Signs are to be displayed to this effect at the entrance to the premises"
- **Provide door supervisors with search wands** "That SIA registered door supervisors shall be provided with a hand held search wand, maintained in good working order, for use during all personal searches of customers, staff and performers"
- Provide search arches "That a search arch shall be maintained in good working order at the entrance to the premises. All admissions and re-admissions to the premises, including staff and performers, will be via the search arch"
- **Develop a drugs policy in conjunction with local police –** "That a drugs policy shall be developed to the satisfaction of the local police licensing office. Signs shall be exhibited at the entrance to the premises stating "Drugs free zone"

- Provide a drugs and weapons drop box "That a secured drugs and weapons drop box shall be provided in an area at the front of the premises under the permanent control of management and security staff. All seized drugs or weapons shall be deposited in the box and handed over to the police in accordance with the police code of practice"
- **Maintain a photographic log of all customers –** "That a full face photograph shall be taken of all customers admitted to the premises and a photographic log maintained for inspection by any authorized officer upon request"
- **Establish a last admissions policy** "That there shall be no new admissions of the public nor re-admissions after ... (state relevant time) ... Signs shall be displayed tot his effect"
- **Establish a dispersals policy –** "That a dispersals policy shall be established aimed at encouraging customers to leave the premises quickly, in a quiet and orderly manner"
- 17 Use text pagers to keep in touch "That text pagers shall be provided to key management personnel with contact details provided to the local police station in order to receive safety and other important messages". Such condition should be supported by additional conditions covering that "Text/paging equipment shall be maintained in good working order at all times"; "The pager link must be activated, made available to and monitored by the DPS or a responsible member of staff at all times that the premises are open tot he public"; "Any police instructions are complied with whenever given" and "All instances of crime or disorder are reported via the text/pager link by the DPS or responsible member of staff to an agreed police contact point".
- **Provide seating for customers –** "That sufficient seating will be provided to ensure that at any time the majority of customers will not have to stand"
- **Provide restrictions on drinking areas –** "That the consumption of alcohol will only be permitted within the areas identified on the plans submitted to the licensing authority (to be appended on the licence document)"
- **Co-operate with the police and local council on venue hire agreements –** "That a completed venue hire agreement form (obtainable from the licensing authority) will be provided with at least one months prior notice to the police and the licensing service, in respect of every external promoter or new internal promoter engaged at the premises" and "That the licensee will comply with instructions not to engage a specified promoter where the police indicate that this is necessary for the purpose of promoting the prevention of crime and disorder licensing objective"
- Avoid certain exhibitions and performances "That no film will be exhibited / performance take place that is likely to stir up hatred or incite violence toward or against any section of the public on the grounds of colour, race or ethnic or national origin, disability, religious beliefs, sexual orientation of gender" and "That no film will be exhibited or performance given that will lead to disorder"
- 22 Use bona fide suppliers "That stocks of alcohol and tobacco shall only be sourced from authorized or bona-fide suppliers"
- 23 Deny access to hawkers "That no counterfeit products, such as alcohol, tobacco products, DVDs, CDs and other goods, may be supplied from the premises either by the staff or by hawkers entering for that purpose"

Cumulative effect

- 144. This authority will not consider the issue of "need" in determining any licence application, as this is a matter for planning control and the market. Thereby, this council will not impose quotas of premises or licences.
- 145. However, the authority recognises that in areas where the number, type and density of premises selling alcohol are high, serious problems of nuisance and disorder may be arising or have begun to arise outside or some distance from licensed premises. Guidance to the Act sets out that the cumulative impact of licensed premises on the promotion of the licensing objectives is a proper matter for a licensing authority to consider within its licensing policy and that local authorities may adopt special local saturation policies. However, there should be an evidential basis for a decision to include a special policy within the statement of licensing policy.
- 146. In summary, the steps to be followed in considering whether to adopt a special policy within the policy are:
 - (a) Identification of concern about crime and disorder or public nuisance;
 - (b) Consideration of whether there is good evidence that crime and disorder or nuisance are happening and are caused by the customers of licensed premises or that the risk of cumulative impact is imminent;
 - (c) Identifying the boundaries of the area where problems are occurring;
 - (d) Consultation with those specified by section 5(3) of the 2003 Act and subject to the outcome of the consultation;
 - (e) Include and publish details of the special policy in the licensing policy statement.
- 147. The effect of adopting a special policy of this kind is to create a rebuttable presumption that applications for new premises licences or club premises certificates or variations that are likely to add to the existing cumulative impact will normally be refused, following relevant representations. This, unless the applicant can demonstrate in their operating schedule, that there will be no negative cumulative impact on one or more of the licensing objectives.
- 148. However, it should be stressed that this presumption does not relieve responsible authorities or interested parties of the need to make a relevant representation, referring to information which had been before the licensing authority when it developed its statement of policy, before a licensing authority may lawfully consider giving effect to its special policy. If there are no representations, the licensing authority must grant the application in terms that are consistent with the operating schedule submitted.
- 149. The Council will not use such policies solely
 - As the grounds for removing a licence when representations are received about problems with existing licensed premises; or
 - To refuse modifications to a licence, except when the modifications are directly relevant to the policy, for example where the application is for a significant increase in the capacity limits.

Local saturation policies

- 150. Since the introduction of the Licensing Act 2003, the council's licensing committee has been monitoring the impact of licensed premises operation across the local Southwark community.
- 151. On 5 November 2008, council assembly agreed that it was appropriate and necessary to introduce a local policy dealing with the cumulative impact of licensed premises within the Camberwell and Peckham areas of the borough. The assembly added a third policy area, covering Borough and Bankside, on 4 November 2009 while also extending the Peckham saturation area. The Borough and Bankside area was itself subject of an extension on 6 April 2011.
- 152. In each case decisions to introduce local policies have been taken having had regard to statistical analysis of recent figures for:
 - Analysis of statistics, prepared in conjunction with the partnership analyst, of alcohol related violence against the person (VAP) and calls to the police regarding disorder and rowdiness associated with licensed premises (CAD);
 - Analysis of complaints received by the council's environmental protection team regarding nuisance associated with licensed premises;
 - Analysis of figures provided by the London ambulance service of alcohol related ambulance pick-ups;
 - Responses to local consultation carried out in accordance with section 5(3) of the Licensing Act 2003; and
 - Submissions made directly to the committee from the local community
- 153. Details of the boundaries of the three saturation areas and also the categories of premises to which the policies apply are set out below. Full information on the current position within each policy area (including up to date analysis of statistics, summary of submissions and a map of each area) is available from the licensing team.
- 154. All applicants for new premises licences or for variation of existing premises licences, in respect of classes of premises affected by the policy, within these areas, must address the saturation concerns within their application and operating schedule. Applicants are advised to discuss their application and how this may be affected by the policy with the licensing team in the first instance.
- 155. While the council notes that Guidance states that "it would not normally be justifiable to adopt a special policy on the basis of a concentration of shops, stores or supermarkets selling alcohol for consumption off the premises" each of the policies incorporates off-licences and similar premises. These decisions have been taken on the basis of concerns relating to issues of under-age sales and proxy purchasing, street drinking and "pre-loading".
- 156. The effects of each policy will continue to be monitored and reported to the licensing committee periodically. A policy will be removed when no longer needed.

Camberwell saturation policy

- 157. The boundary of the area is defined as follows From Camberwell New Road at the junction with Wyndham Road progressing via Camberwell Road / Bowyer Place / Edmund Street / Benhill Road / Wilson Road / Graces Road / Graces Mews / Camberwell Grove (via alley) / Grove Lane / De Crispgny Park / Denmark Hill following the Lambeth boundary to Coldharbour Lane / Denmark Road / Flodden Road and Camberwell New Road returning to the start.
- 158. The classes of premises to which the policy applies is defined as follows night clubs; pubs and bars; off-licences, grocers, supermarkets, convenience stores and similar premises

Peckham saturation area

- 159. The boundary of the area is defined as follows Commencing on Peckham High Street at the junction with Kelly Avenue progressing to Gatonby Street / Sumner Road / Jocelyn Street / cross Peckham Hill Street / Goldsmith Road / Meeting House Lane / Consort Road / Heaton Road / Sternhall Lane / McDermott Road / Maxted Road / Bellenden Road / Chadwick Road / Lyndhurst Way / Peckham Road to junction with Talfourd Road and back to start.
- 160. The classes of premises to which the policy applies is defined as follows night clubs; pubs and bars; off-licences, grocers, supermarkets, convenience stores and similar premises

Borough and Bankside saturation area

- 161. The boundary of the area to which the policy applies is defined as follows Commencing at the Lambeth border on the Thames and moving southward to Waterloo Road then St George's Circus / Borough Road / Borough High Street / Long Lane / Crosby Row / Snowsfields / Bermondsey Street directly north to the river frontage and then westward back to the Lambeth border.
- 162. The classes of premises within the area to which the policy shall apply will be night-clubs / public houses & bars / restaurants & cafes / off-licences, supermarkets and grocers.

Licensing hours

- 163. This authority recognises the Government's strong belief that, prior to the introduction of the Licensing Act 2003, fixed and artificially early closing times were one of the key causes of rapid binge drinking prior to closing times; and one of the causes of disorder and disturbance when large numbers of customers were required to leave premises simultaneously.
- 164. It is understood that the aim, through the promotion of the licensing objectives, should be to reduce the potential for concentrations and achieve a slower dispersal of people from licensed premises through flexible opening times. However, this authority notes that the latest Guidance to the Act states that there is no presumption in favour of lengthening licensing hours and the four licensing objectives should be paramount considerations at all times. Thereby, while applications will be considered upon their own merits, in relevant cases where objections are received to an application and the council's licensing committee believes that granting the hours sought would undermine the licensing objectives,

the committee may reject the application or grant it with appropriate conditions and / or different hours from those requested. Furthermore, relevant saturation policies will be borne in mind where applicable. This authority recognises that all areas are not the same and decisions on opening times must be taken accordingly.

- 165. Where applications relate to premises that are located within more densely populated areas, stricter controls with regard to noise control may be necessary. For guidance please see section 9 on the prevention of public nuisance.
- 166. Shops, stores and supermarkets selling alcohol will normally be permitted to provide sales of alcohol for consumption off the premises, at any time when the retail outlet is open for shopping. However, a restriction on the hours during which alcohol sales may be permitted will be considered where relevant representations are received and there are good reasons, based on the licensing objectives, for limiting those hours.

Designated premises supervisor / personal licence holders and authorisations for the retail sale of alcohol

- 167. This authority recognises that neither the Act nor the mandatory conditions outlined elsewhere in this chapter require either a designated premises supervisor or any other personal licence holder to be present on the premises at all times when alcohol is sold or supplied. Nor does the fact that every sale or supply of alcohol must be made under the authority of a personal licence holder mean that only personal licence holders can make such sale or supply or that they must be personally present at every transaction.
- 168. However, this authority wishes to emphasise that the designated premises supervisor and the premises licence holder remain responsible for the premises at all times including compliance with the terms of the Licensing Act and conditions attached to the premises licence to promote the licensing objectives.
- 169. This authority considers it reasonable to expect therefore that either the designated premises supervisor or another personal licence holder will normally be present upon the premises at all times that alcohol is either sold or supplied or that proper arrangements are in place for authorisation of staff in their absence and for their monitoring. This authority does recognise that there may be occasions when it is not possible for the designated premises supervisor or another personal licence holder to be present but this authority considers that this should not occur for any extended period and emphasises that there can be no abdication of responsibility in their absence.
- 170. Furthermore, this authority expects that authorisations of other staff to sell and supply alcohol made by the designated premises supervisor and any other personal licence holders must be meaningful and be properly carried out.
- 171. It is understood that, ultimately, whether or not an authorisation has been given within the meaning of the Act will be a matter for a court to determine on the evidence before it when the issue arose, but this authority notes the Secretary of State considers that the following factors should be relevant in considering whether there was real authorisation:

- a) The person(s) authorised to sell alcohol at any particular premises should be clearly identified;
- b) The authorisation should have specified the acts which may be carried out by the person being authorised;
- c) There should be an overt act of authorisation, for example a specific written statement given to the individual(s) being authorised; and
- d) There should be in place sensible arrangements for the personal licence holder to monitor the activity that they have authorised on a reasonably regular basis.
- 172. Attention is drawn; however, to the fact that recent provision has been made for certain community premises to apply for the removal of the mandatory condition relating to the requirement for a DPS and personal licence and replace this with an alternative condition making a management committee or board of individuals responsible for the supervision and authorisation of alcohol sales. Where such application is made the council will expect the applicant to demonstrate that the arrangements for the management of the premises by the committee or board are adequate to ensure adequate supervision of the supply of alcohol on the premises. In cases where it is intended to make the premises concerned available for private hire, this authority will wish to consider the arrangements for hiring agreements. This consideration will include steps to be taken to ensure that the hirer is aware of the licensing objectives and offences in the 2003 Act and to ensure that no offences are committed during the period of the hire.

Door Supervision

173. Door supervisors will often play an important key role in ensuring that well managed premises are free from crime and disorder and nuisance problems. A reminder is given that where a premises licence (other than for plays or films) includes a condition that at specified times there must be persons employed for the purposes of security, protection, screening the suitability of persons entering premises or dealing with conflict in pubs, clubs and other licensed premises open to the public, then such persons must be registered with the Security Industry Authority (SIA). In any event it is in operator's own interests to ensure that all security staff employed, whether under contract or employed in-house are so licensed, since the operator may be legally liable for their actions. Licensed door supervisors will have undergone an identity and criminal record check and have received suitable training. More information on the SIA is available from their website at www.the-sia.org.uk.

Police promotion event risk assessment form 696

- 174. This authority recommends that for significant events (see note below) further specific comprehensive risk assessments are undertaken by premises licence holders to identify and minimise any risk of serious violent crime. Accordingly, for premises that wish to stage promotions, or events (as defined below) this authority recommends that applicants address the risk assessment and debrief processes in their operating schedules.
- 175. This authority further recommends the metropolitan police promotion / event risk assessment form 696 and the after promotion / event debrief risk assessment form 696a (both available from the local police licensing office) as useful and effective tools for this purpose. Where the risk assessment forms are used to assess the

likely risks from any promotion or event, this licensing authority anticipates that these will be completed in consultation with the metropolitan police. Risk assessments should be submitted to the metropolitan police and the licensing authority within 14 days of any proposed event and debrief forms submitted within 14 days of the conclusion.

176. Note: The metropolitan police define "a significant event" as any occasion in a premises licensed under the provisions of the licensing act 2003, that is promoted / advertised to the public at any time before the event; predominantly features DJs or MCs performing to a recorded backing track; runs anytime between the hours of 10pm and 4am; and is in a nightclub or large public house. Licensees should consult with the local police licensing office if in any doubt as to whether an event is "significant".

Entertainment involving striptease and nudity

- 177. On 20 October 2010, council assembly resolved to adopt the provisions under Schedule 3 to the Local Government (Miscellaneous Provisions) Act 1982 as amended by section 27 of the Policing and Crime Act 2009, with effect from 1 April 2011. This legislation established a new extended licensing regime for sex establishments, covering sex shops, sex cinemas and sex entertainment venues.
- 178. From 1 April 2011, any person wishing to operate a lap dancing club or similar in the Southwark area will require to be licensed as a sex entertainments venue under a sex establishments licence, obtained from this authority. The exemption that existed in law that enabled lap dancing clubs or similar to operate under the provisions of the 2003 Act no longer apply, excepting that occasional striptease performances are permitted without a sex establishments licence. Occasional performances are limited to no more than 12 per year, each of less than 24 hours duration and each at least one month apart.
- 179. This authority's approach to the licensing of sex establishments is covered in full under the separate Southwark Sex Establishments Licensing Policy, which is available from the licensing section.
- 180. Beyond the remit of the 1982 Act (as amended) where activities specified in the operating schedule include any permitted adult entertainment (involving full or partial nudity, occasional striptease or any other kind of adult entertainment) the licensing authority will give particular consideration to the implications of the proposals across the four licensing objectives.
- 181. In such circumstances, this authority advises that applicants should give specific consideration when conducting their risk assessments and operating schedules to the appropriate management controls. This must include issues around the protection of children and young people from harm.
- 182. Where occasional striptease performances are provided, the licensing authority will normally expect the operating schedule and, if the application is granted, the licence conditions to address all relevant matters in pursuit of the licensing objectives. These are likely to include conditions relating to:

- The exclusion of persons under 18 at all times from the premises and the prevention of views into the premises;
- The prohibition of exterior advertising of the sex-related entertainment at the premises;
- The prohibition of leafleting or touting for business;
- That the provision of services will be to seated customers only;
- The prohibition of the participation of customers in the performance;
- The maintenance of a minimum distance of 1 metre between performer and customers during the performance;
- The provision of CCTV and the maintenance of a library of recordings;
- A code of conduct for dancers including appropriate disciplinary procedures for failure to comply with the code;
- Rules of conduct for customers, including appropriate procedures for breach of these rules:
- Procedures to ensure that all staff employed in the premises have preemployment checks including suitable proof of identity, age and (where required) permission to work;
- The provision of supervisors and security staff; and
- Other necessary steps to be taken, including in the recruitment or supply of performers, is free of the influence or organised criminal activity.

The sale and possession of smuggled, bootlegged or counterfeit products and the handling of stolen goods

- 183. The guidance to the Act indicates that offences can be committed by certain persons in licensed premises if they allow the sale or supply of smuggled, bootlegged or counterfeit goods on the premises. This is in particular relation to alcohol or tobacco products, but may also include the sale of other counterfeited products such as DVDs and CDs and the handling of stolen goods.
 - Smuggled foreign brands illegally brought into the UK;
 - Bootlegged (duty free) foreign products brought into the UK in illegal quantities and / or resold in the UK; and
 - Counterfeit Fake goods and packaging. Cheap low quality and often dangerous products packaged to look like premium UK brands.
- 184. Offences are for selling illicit products but also for employees permitting sales and for employees buying or selling. To prevent these offences occurring, the licensee should consider taking certain precautions. Advice on this can be obtained from the council's trading standards service.

Theft of personal property

- 185. Recent analysis of crime statistics has given rise to increasing concern over levels of theft of personal property within licensed premises, especially within the developing northern-western riverside area of the borough. Licence applicants are asked to give consideration to appropriate control measures within their risk-assessments and operating schedules.
- 186. An important first step is to ensure that measures are taken to raise the level of awareness among customers of the risk of theft of personal property. This is especially important within premises where there is a high level of transient, tourist

trade. Other potential controls measures might include controlled cloakroom areas; provision of Chelsea clips under tables; ensuring all areas of the premises are well lit and able to be observed by staff; provision of internal CCTV; ensuring adequate staffing levels and providing appropriate staff training. Officers from the licensing and / or police teams will be happy to visit and discuss arrangements at your premises.

SECTION 8 – THE SECOND LICENSING OBJECTIVE – ENSURING PUBLIC SAFETY

Note: For how this policy is intended to apply see Section 5.

- 187. The act covers a wide range of premises that require licensing, including cinemas, concert halls, theatres, public houses, stores, cafes / restaurants and fast food outlets. Each of these premises presents a mixture of risks. While many are common to most premises, others are unique to specific operations. It is essential that premises are constructed or adapted and managed in a manner which acknowledges these risks and safeguards occupants against them.
- 188. This authority is committed to ensuring the safety of people visiting and working in licensed premises. Thereby, this authority expects applicants for premises licenses to demonstrate in their operating schedules that suitable and sufficient steps are to be taken to ensure the licensing objective of public safety in and around the vicinity of their premises.
- 189. Operating schedules accompanying premises licence and club premises certificate applications should be prepared on the basis of a risk-assessment of the potential risks and hazards posed to the public under the premises operation.
- 190. Applicants are recommended to seek advice from the council's occupational health and safety officers and / or local fire safety officers (see contact details in section 12 of this document) when carrying out their risk assessments and preparing their operating schedules.
- 191. The following section of this statement of policy is given over to providing a helpful broad guide to applicants as to the range of issues they may wish to consider under their risk-assessments.
- 192. It is emphasised that this is neither meant to be a definitive list nor will all the issues be relevant in every circumstance. It is recognised that applicants know their premises and method of operation and controls proposed under the operating schedule should properly reflect the proposed operation of the premises concerned.
- 193. While, as far as possible, this policy is not intended to duplicate existing legislation and other regulatory regimes that already place obligations on employers and operators, and that it is recognised that there is no requirement in the 2003 Act to do so, applicants may find that it helps the licensing process to provide a copy of the completed risk-assessment upon which the operating schedule is based to the relevant responsible bodies for consideration at the time of application.

Other relevant legislation

194. Applicants are firstly reminded of the general duties imposed under other legislation (i.e. the Regulatory Reform (Fire Safety) Order 2005) and the Health and Safety at Work Regulations 1999.

Regulatory Reform (Fire Safety) Order 2005

- 195. Under the Regulatory Reform (Fire Safety) Order 2005 the "responsible person" as defined by the order, must
 - Take such general fire precautions as will ensure, so far as is reasonably practicable, the safety of his or her employees;
 - Take such general fire precautions as may be reasonably required in the circumstances of the case to ensure that the premises are safe for the use by other persons;
 - Make a suitable and sufficient assessment of fire risks to which relevant persons may be exposed for the purposes of identifying the general fire precautions he or she needs to take; and
 - The responsible person MUST record the information gathered by the risk assessment for all premises covered by the Licensing Act 2003, in particular, the significant findings; the measures taken, or to be taken, to ensure general fire safety; and any person identified as being especially at risk.
- 196. This authority recognises that it may not impose any condition relating to any requirement or prohibition that is or could be imposed under the "Fire Safety Order" and thereby that conditions relating purely to fire safety matters may not be placed on premises licences. Further information and guidance about the Order and fire safety legislation is available from the Communities and Local Government website at www.communities.gov.uk/fire.

Health and Safety at Work Regulations 1999

- 197. The Health and Safety at Work Regulations 1999 state that:
 - (1) Every employer shall make a suitable and sufficient assessment of
 - (a) The risks to the health and safety of his employees to which they are exposed whilst they are at work; and
 - (b) The risks to the health and safety of persons not in his employment arising out of or in connection with the conduct by him of his undertaking, for the purpose of identifying the measures he needs to take to comply with the requirements and prohibitions imposed upon him by or under the relevant statutory provisions.
 - (2) Every self-employed person shall make a suitable and sufficient assessment of -
 - (a) The risks to his own health and safety to which he is exposed whilst he is at work; and
 - (b) The risks to the health and safety of persons not in his employment arising out of or in connection with the conduct by him of his undertaking,
- 198. The Health and Safety Executive helpfully provides a range of example risk-assessments which may be viewed at http://www.hse.gov.uk/risk/casestudies/.

The examples provided include for night-clubs, pubs, village halls, off-licences and convenience stores.

A broad guide to risk-assessment

- 199. It is not possible to provide a complete list of the matters which might be covered by risk assessments since circumstances vary from case to case. However, depending on the circumstances of the case and the outcome of the risk assessment, applicants should consider whether the matters indicated on table 3 (on the following page) should be addressed in their operating schedule:
- 200. It should also be understood that it is not necessarily a simple process to convert the operation of a premises to another usage or activity. For instance, the addition of music and dancing at premises that previously only dealt with the sale and supply of alcohol might raise a number of issues. For instance the provision of a dance floor should be properly risk-assessed to provide customer comfort and a safe environment. Adequate supervision of dancers and a safe position for the DJ or band might be necessary, while issues of lighting, special effects, potential for noise and ancillary issues as bottle storage / collection might all arise.
- 201. Those preparing operating schedules are recommended to refer to the following guidance:
 - Model National and Standard Conditions for Places of Public Entertainment and Associated Guidance ISBN 904031 11 0 (Entertainment Technology Press – ABTT Publications);
 - The Event Safety Guide A guide to health, safety and welfare at music and similar events (HSE 2002) ("The Purple Book") ISBN 0 7176 2453 6;
 - Managing Crowds Safely (HSE 2000) ISBN 0 7176 1834 X
 - 5 Steps to Risk Assessment: Case Studies (HSE 1998) ISBN 07176 15804;
 - The Guide to Safety at Sports Grounds (The Stationery Office 1997) ("The Green Guide") ISBN 0 11 300095 2;

 - Fire Safety Risk Assessment Open Air Events and Venues" (ISBN 978 1 85112 823 5) which is available from the Communities and Local Government website www.communities.gov.uk/fire;
 - The London District Surveyors Association's "Technical Standards for Places of Public Entertainment" ISBN 0 9531229 2 1;
 - Safer Clubbing Produced by the Home Office in conjunction with the London Drugs Policy Forum
- 202. The following British Standards should also be considered:
 - BS 5588 Part 6 (regarding places of assembly);
 - BS 5588 Part 9 (regarding ventilation and air conditioning systems);
 - BS5588 Part 9 (regarding means of escape for disabled people);
 - BS 5839 (fire detection, fire alarm systems and buildings); and

• BS 5266 (emergency lighting systems)

Table 3 – Broad guide to public safety considerations			
Site and general	Site and layout arrangements / access for emergency		
arrangements	vehicles / entry arrangements / pre-opening safety checks		
Means of escape	Horizontal escape / seating and gangways / vertical escape		
	/ arrangement and maintenance of final exits and escape		
	routes to exits / escape signage		
Structure	Stability / barriers and guarding / temporary structures		
Fire safety	Fire separation / fire resistance and finishes / smoke		
,	ventilation and containment / safety curtains / hangings,		
	drapes, decorations and upholstery / lightning protection /		
	arrangements for contacting emergency services		
Building services	Electrical installations, including temporary electrical		
	installations / lighting, including emergency lighting /		
	ventilation / heating and fuel / water supplies and drainage		
Hygiene	Sanitary accommodation / cloakrooms and changing rooms		
	/ drinking water supplies / refuse storage and disposal		
Fire and emergency	Fire and emergency warning systems / fire extinguishing		
systems	arrangements / generators		
Communications	Communications systems / safety signs and notices		
Special installations	Mechanical installations / stage engineering / special effects		
and effects	/ hearing protection for staff / hazardous materials.		
	Note: The operating schedule should identify all special effects to be used on the premises such as flame, lasers, pyrotechnics, smoke, fog, foam or firearms. A risk assessment should be carried out in respect of each and should detail steps to be taken to prevent risks to patrons and performers.		
Capacity	The maximum capacity of the premises and (as appropriate) the maximum capacity for separate areas within the premises / procedures to record the number of persons entering and leaving the premises		
Staffing provision	Note: The LFEPA request that the proposed maximum capacity is submitted with the scale (1:100) plans, the LFEPA will respond either agreeing with the proposal or recommending an alternative capacity. Proof that, where necessary, SIA registered door staff are		
and training	employed / Details of steps that have been taken or will be		
3.13 3.311119	taken to train staff in matters relating to public safety / first		
	aid provision		
Access and egress	Details of access and egress provision for disabled persons		
for disabled persons	should be provided. Particular emphasis should be placed		
	on evacuation procedures.		
L			

203. Where its discretion is engaged following the consideration of relevant representations the Licensing Authority will consider attaching relevant conditions to licences and permissions to promote public safety where these are not provided

for within other legislation and are necessary for the promotion of that licensing objective.

Safe capacities

- 204. The Guidance to the Act sets out that "safe capacities" should only be imposed where necessary for the promotion of public safety or the prevention of disorder on the relevant premises.
- 205. The position of the authority and the responsible authorities is that the importance of the role of safe capacities in terms of both public safety and crime and disorder should not be overlooked.
- 206. Accordingly it is recommended that wherever the setting of a safe capacity will assist issues of crowd safety or public order that this is addressed by the operating schedule. It should be borne in mind that the safe capacity should be that which applies at any material time that licensable activities may take place under the premises licence.
- 207. It should also be noted that where an applicant wishes to take advantage of the special provisions in Section 177 of the Act relating to small venues then a capacity must be set in consultation with the fire authority.
- 208. For guidance in calculating safe capacities applicants may refer to the London District Surveyors Associations Technical Regulations referenced in this Section.

A proactive approach to customer safety

- 209. Applicants are also encouraged to take a proactive approach to general safety issues, including that of their customers, and to discuss appropriate measures with the responsible authorities concerned. Arrangements that might be taken into account could include steps;
 - To monitor that customers present upon the premises are not continued to be supplied with alcohol to a point where they become drunk;
 - To ensure that customers are able to return home safely, for instance, by providing information on local transport availability or making arrangements with reputable cab companies;
 - To ensure against drinks being "spiked" or otherwise tampered with; and
 - To ensure all (internal and external) parts of the premises are adequately lit and may be monitored by staff.

SECTION 9 – THE THIRD LICENSING OBJECTIVE – THE PREVENTION OF PUBLIC NUISANCE

Note: For how this policy is intended to apply see Section 5

- 210. Licensed premises, especially those operating late at night and in the early hours of the morning, can cause a range of nuisances impacting on people living, working or sleeping in the vicinity of the premises. Principal concerns relate to noise nuisance, light pollution and noxious smells.
- 211. This authority as a result of the above, will expect applicants for premises licences and club premises certificates to demonstrate in their operating schedules that suitable and sufficient steps are to be taken to prevent nuisance in and around the vicinity of their premises and events.
- 212. Operating schedules accompanying applications should be prepared on the basis of a risk-assessment of the potential sources of nuisance Relevant issues might include:
 - Preventing noise and vibration escaping from the premises, including music, noise from ventilation equipment and human voices;
 - Preventing disturbance by customers arriving at, gathering outside of, being admitted or re-admitted to or departing from the premises, particularly between 2300 and 0700 hours;
 - Preventing customer or vehicle queues forming outside of the premises, or where some form of queuing is necessary, the steps to prevent disturbance or obstruction:
 - Ensuring clear up operations conducted by staff do not cause a nuisance and that staff leave the premises quietly;
 - Addressing arrangements made for parking by patrons
 - Considering whether there is sufficient public transport provision and where licensed taxis or private hire vehicles are likely, arrangements made to prevent disturbance to local residents
 - Controlling disturbance that may be caused by the use of gardens and other open-air areas including the highway – particularly in relation to the recent smoking ban and passage of patrons between customers and outside areas
 - Preventing nuisance from the positioning and operation of plant and machinery
 - Restricting delivery and collection times
 - Limiting nuisance caused by the positioning of external lighting, including security lighting
 - Preventing refuse storage and disposal problems and the accumulation of litter in the vicinity of the premises
 - The need for regular patrols of the curtilage of the premises

213. Applicants are advised to refer to:

• The Institute of Acoustics "Good Practice Guide on the Control of Noise from Pubs and Clubs";

- Noise at Work Guidance for employers on the Control of Noise at Work Regulations 2005 issued by the Health and Safety Executive and obtainable from http://www.hse.gov.uk/pubns/indg362.pdf;
- In the case of large music events, the Code of Practice on Environmental Noise Control at Concerts produced by the Noise Council;
- Guidance on the Control of Odour and Noise from Commercial Kitchen Exhaust Systems produced by DEFRA; and
- Guidance Notes for the Reduction of Obtrusive Light Institute of Lighting Engineers (2005)
- 214. Applicants are recommended to seek advice from the council's environmental protection team when carrying out their risk assessments and preparing their operating plans and schedules.
- 215. For this purpose, table 4 on the following page, is provided as a helpful guide to applicants as to the types of controls that may be considered when assessing the steps necessary for the promotion of the prevention of nuisance licensing objective, following receipt of relevant representations. It should be noted that it is likely that additional controls may be necessary to prevent nuisance after 2300 hours.
- 216. This is neither meant to be a definitive list of possible preventative measures nor will the controls listed be necessary in every circumstance. It is recognised that applicants know their premises and method of operation and controls proposed under the operating schedule should properly reflect the proposed operation of the premises concerned.
- 217. While it is recognised that there is no requirement in the 2003 Act to do so, applicants may find that it helps the licensing process to provide a copy of the completed risk-assessment upon which the operating schedule is based to the relevant responsible bodies for consideration at the time of application.
- 218. Where its discretion is engaged following consideration of relevant representations, the licensing authority will consider attaching appropriate conditions to licences and permissions to prevent nuisance.

Outdoor events

- 219. Each year Southwark hosts a large number of outdoor community events. These events add to the vibrancy of the borough and are enjoyed by both residents and visitors alike. It is important that such events are organised and managed so as to ensure that minimum disturbance is caused to people living and working nearby. To help achieve this applicants are asked to submit an **Event Management Plan** to the Council's Environmental Protection Team with their application.
- 220. The Environmental Protection Team, as a Responsible Authority, will assess all licence applications for their impact with regard to public nuisance. Nuisance from outdoor events can be created by many activities, for example:
 - Set up rigging / de-rigging
 - Plant & machinery

Tab	Table 4 – Broad guide to sound containment and reduction measures				
1	The provision of acoustic double door lobbies				
2	The use of double glazing / sound resistant glass				
3	The use of acoustic seals on doors and windows				
4	Keeping windows and doors closed during performances of licensable entertainment				
5	The use of acoustic baffling to the ventilation extract and intake system				
6	The provision of additional air conditioning / air cooling				
7	The provision of sound transmission insulation to party walls				
8	Mounting of sound equipment brackets and free standing equipment away from party walls / floors and use of isolation material				
9	The provision of alarms to fire doors / other private external doors				
10	Management control of entrance / exit lobby to ensure integrity maintained				
11	Restrictions on live music / amplified sound being played in the outside area / garden / patio / marquee or external speaker connection to internal speakers				
12	The installation and use of a sound-limiting device for all amplified sound				
13	The connection of the fire doors / private external doors to the sound limiting device				
14	The reduction of the volume or ceasing of the music / amplified sound e.g. 30				
	minutes prior to closure				
15	Management of customers queuing to gain entry to the premises				
16	Management of periodic perimeter checks				
17	Management of exit and re-entry smokers / limit the numbers of smokers in the designated area				
18	The use of management / security staff to supervise external areas used for consumption				
19	Restriction of the use of the outside / garden / patio area after 2300 hours				
20	The use of notices / announcements requesting customers to leave the area quietly				
21	Arrangements with local car-hire to enter premises to collect customers				
22	Notices advising on the use of horns, loud music or slamming doors from private				
	vehicles outside the premises and in ancillary car parks				
23	The quick dispersal of customers from outside the premises after closing time				
24	Restriction on hours for handling and removal of waste, musical equipment and				
	delivery of goods e.g. to between 0800 and 2000				
25	Measures to prevent littering and to clear up any litter e.g. litter patrols				
26	Measures to restrict exterior artificial lighting giving rise to nuisance				
27	Measures to prevent cooking odours giving rise to nuisance				
28	No drinks to be taken outside whilst smoking				
29	Provide a 'drinks safe' area whilst the smoker is outside				

- Vehicle movements
- Music
- People
- Fireworks and special effects
 Odour and smoke from food preparation
- Litter
- Lighting

- 221. It is recommended that event organisers include a **Nuisance Management Section** within the **Event Management Plan** assessing all nuisance-causing activities and setting out the measures proposed to mitigate against any nuisance being caused.
- 222. The nuisance management section should include the following information:
 - Location & site plan
 - Information on the type of event
 - Orientation of stage or music source
 - Access onto & from site and site routes
 - Location of operational plant & vehicles
 - Background noise survey and predictive noise assessment
 - During performance sound management responsibility and recording of levels, complaint response and records of action, post event audit meeting with authority
 - Contact details of a person at the event responsible for liaison with the Council Officers
 - Copy of the leaflet to be dropped to occupiers of nearby noise sensitive premises
 - Location of light sources & light impact assessment
 - Litter control & recycling
 - Noise control measures
 - Maximum noise levels
 - Selection of equipment etc.

Nuisance outside of premises

- 223. Nuisance caused by patrons outside of premises, whether this is from a beer garden, on terraces or on the pavement, is of considerable concern to local residents who live in close proximity to licensed premises. Of course noise is often the main concern, however, obstruction of the highway that prevents people passing by on the pavement, access by emergency services, or street cleansing are other serious concerns.
- 224. Local residents may also be affected by smoke from barbeques and other cooking equipment used in the open air. Licensees should consider carefully the steps required to prevent such nuisances in their operating schedule and in the day to day management of their premises.
- 225. Licensees wishing to place tables and chairs on the public highway should also note that an additional consent and licence is required from the Council's Street Trading section. Contact details are provided in Section 12 of this document.

Smoking and licensed premises

226. The Government's smoke free workplaces initiative came into effect on 1 July 2007. One of its impacts has been to prohibit customers and staff from smoking within licensed premises and enclosed areas. While this constitutes a welcome measure in reducing risk to public health from the affects of second hand smoke, this has also given rise to many instances of nuisance being caused outside of

- licensed premises as licensees have sought to make provision for customers who wish to continue smoking outside of the premises.
- 227. It should be understood that licensees have a responsibility for the conduct of their customers while they are in and around their premises. If provision is made for customers to smoke outside of premises then sufficient management controls must be put into place to ensure that no nuisance or disturbance is caused to local residents. Failure to control customers outside of premises may result in licence reviews being submitted and impact upon the premises licence.
- 228. Licensees are advised to consider the impacts on the local community of any provision made for customers outside of the premises. In cases where the premises is located in a residential area, it may be necessary to consider setting times after which external areas will not be used; to prevent drinks being taken outside of the premises while customers go outside to smoke; to set a no admissions or re-admissions policy in place at a particular time to prevent nuisance; or consider how littering from cigarette butts may be controlled.
- 229. Licensees should be aware that a range of other consents may be necessary to provide smoking shelters, awnings, tables and chairs or space heaters. Please discuss your proposals with the licensing team in the first instance (see contacts section of this policy).

SECTION 10 – THE FOURTH LICENSING OBJECTIVE – THE PROTECTION OF CHILDREN FROM HARM

Note: For how this policy is intended to apply see Section 5

- 230. The licensing objective of the protection of children from harm includes the protection of children from moral, psychological and physical harm.
- 231. The council will expect applicants for premises licences and club premises certificates to demonstrate in their operating schedule that suitable and sufficient steps are to be taken to ensure the licensing objective of protecting children from harm.
- 232. Operating schedules accompanying licence applications should be prepared on the basis of a risk assessment of the potential sources of harm to children. Relevant issues might include;
 - Preventing access to alcohol
 - Removing encouragement for children to consume alcohol
 - Preventing exposure to gaming
 - Preventing exposure to adult entertainments or facilities
- 233. This Council recognises the Southwark Safeguarding Children Board (SSCB) as being competent to provide advice on matters relating to the protection of children from harm. Applicants are advised to seek advice from the Board when carrying out their risk assessments and preparing their operating plans and schedules. Southwark Trading Standards are also available to give advice around preventing under-age sales. See section 12 for contact details.
- 234. This section of the policy offers best practice guidance across some of these issues.
- 235. It is emphasised that this guidance is neither meant to cover all issues nor will the suggestions made be necessary in every circumstance. It is recognised that applicants know their premises and method of operation and controls proposed under the operating schedule should properly reflect the proposed operation of the premises concerned.
- 236. While it is recognised that there is no requirement in the 2003 act to do so, applicants may find that it helps the licensing process to provide a copy of the risk-assessment relating to the protection of children from harm to the SSCB and the trading standards team when submitting the application.
- 237. Where its discretion is engaged following the consideration of relevant representations, the licensing authority will consider attaching conditions to protect children from harm.
- 238. Applicants proposing activity that includes adult entertainment are also referred to the policy set out at pages 41-42. The victims of "people trafficking" often include

young females under 18, and this is a further reason why the policy makes reference to it.

Access to licensed premises

- 239. The act makes it an offence to permit children under the age of 16, who are not accompanied by an adult to be present on premises being used exclusively or primarily for the supply of alcohol for consumption on those premises, under the authorisation of a premises licence, club premises certificate or where that activity is carried on under the authority of a temporary event notice. In addition, it is an offence to permit children under 16 who are not accompanied by an adult, between midnight and 5.a.m. at other premises supplying alcohol for consumption on the premises, under the authority of any premises licence, club premises certificate or temporary event notice.
- 240. The council recognises, however, that licences may be sought for a great variety of premises, including theatres, cinemas, restaurants, concert halls, take-away and fast food outlets as well as public houses and nightclubs.
- 241. It is not the council's intention that the access of children should be any further limited to licensed premises, unless it is necessary for the prevention of harm to children. Nor is it the council's intention to require that access to any premises is given at all times. While the council would encourage applicants to make provision for the access of children this remains at the applicant's discretion.
- 242. However, this authority notes that the Secretary of State has recommended that following receipt of relevant representations (unless there are circumstances justifying the contrary) there should be a strong presumption against permitting any access at all for children under 18 years in any premises with evidence of known associations with or likely to give rise to:
 - Heavy or binge or under-age drinking;
 - Drugs;
 - Significant gambling; or
 - Any activity of entertainment (whether regulated entertainment or not) of a clearly adult or sexual nature
- 243. Additionally, the Secretary of State has said that for any premises, not serving alcohol for consumption on the premises, but where the public are allowed on the premises after 11.00.p.m. in the evening, following receipt of relevant representations, there should be a presumption against the presence of children under the age of 12 unaccompanied by adults after that time.

The purchase and consumption of alcohol by minors

- 244. The council shares the government's concern over the purchase and consumption of alcohol by minors.
- 245. The NHS report on 'Smoking, Drinking and Drug Use Among Young People in England in 2008' notes the average weekly intake for pupils aged 11-15 who had drunk alcohol in the week before they were surveyed was 14.6 units. More than double since 1990. Furthermore, 12,718 children aged between 11-17 were

- admitted to hospital in 2008-9 with an alcohol related condition (3,556 aged 11-15 and 9,164 aged 16-17). The UK has one of the highest rates in the EU of admissions to hospital or A&E due to alcohol use by 15-16 year olds.
- 246. Frequency of drinking is associated with offending in children and young people. The 2004 Offending, Crime and Justice Survey found those who drank alcohol once a week or more committed a disproportionate volume of crime, accounting for 37% of all offences but only 14% of respondents.
- 247. In the years since the 2003 Act came into being additional enforcement programmes have been funded through the neighbourhood renewal programmes, aimed at reducing the level of under-age sales within the borough. This was in addition to funding made available across the UK through the Home Office's own alcohol harm reduction programme.
- 248. Figure 5 below shows the results of alcohol test purchase results within Southwark since the 2003 Act was introduced.

Figure 5 – Alcohol test purchase results				
Year	Test purchases	Compliance (no. of sales)	Compliance	
			rate %	
2005 - 2006	146	65	45%	
2006 – 2007	102	76	75%	
2007 – 2008	78	57	73%	
2008 – 2009	123	101	82%	
2009 - 2010	41	29	70%	

- 249. As noted elsewhere in this document, since October 2010 it has been a standard requirement of all premises licences which authorise alcohol sales, that the licence holder must have an age verification policy in place.
- 250. The trading standards service, in it's role as responsible authority, considers each individual licence application received by the licensing authority and makes recommendations on the appropriate range of management controls around this issue. A broad guide to the service's recommended approach to controlling underage sales is provided in table 6. While this is intended as guidance only, this authority recommends that all operators should give careful consideration to introducing these controls into their management practices and including them within their operating schedules. We have found that the rate of compliance has been assisted by take-up of these controls. The recommendations support the national 'Think 25' policy.
- 251. Trading standards also provide practical support. This includes a proof of age card; a 'Good Trader' award scheme; a range of business advice information leaflets; and other support materials such as posters and point of sale material. The service also offers an in-house training programme which, in the near future, will also be backed by the Trading Standards Institute sponsored 'Fair Trader' award.

Table 6 - Broad guide to controlling under-age sales measures		
Signage	That notices shall be displayed informing customers and staff that sales of alcohol will not be made to under 18s and that age verification will be required	
Proof of age compliance scheme	That an age verification scheme shall be established and maintained. The scheme shall require the production of evidence of age (comprising any Proof of Age Standards Scheme (PASS) accredited card or passport or driving licence) from any person appearing to staff engaged in selling or supplying alcohol to be under the age of 18 and who is attempting to buy alcohol. Evidence of this scheme (in the form of documented procedures) shall be made available for inspection by authorised officers.	
	or	
	That "Challenge 25" or "Think 25" as part of an age verification scheme shall be supported. The scheme shall require the production of evidence of age (comprising any PASS accredited card or passport or driving licence) from any person appearing to staff engaged in selling or supplying alcohol to be under the age of 25 and who is attempting to buy alcohol. "Think 25" or "Challenge 25" notices shall be displayed informing customers and staff that sales of alcohol will not be made to anyone appearing to be under 25 unless they can prove they are 18 or over. Evidence of this scheme (in the form of documented procedures) shall be made available for inspection by authorised officers.	
Staff	That all staff involved in the sale of alcohol shall be trained in the age	
training	verification scheme and records of such training shall be retained on the premises and made available for inspection by authorised officers.	
Incident log	That an incident log shall be maintained and details of all age related refusals recorded. This book shall be reviewed monthly by the DPS and any actions taken recorded in the book and signed off by the DPS. This log shall be retained on the premises and made available for inspection by authorised officers.	
CCTV	That, if a CCTV recording system is installed inside the premises, it shall be positioned to capture the sale of alcohol. A library of recordings taken by the system shall be maintained for 31 days and made available to authorised officers upon request.	
Presence of personal licence holder	That a personal licence holder shall be on the premises at all times that alcohol is supplied.	
EPOS System	That, an EPOS (electronic point of sale) system is installed at the premises, it shall be configured to prompt the operator to check proof of age when an alcoholic product is scanned. This prompt should require the operator to confirm the purchaser is over 18 before the next item can be scanned or the transaction completed.	
Monitoring of displays	That the DPS shall ensure that, as far as is reasonably practical, alcohol is displayed in an area which can be constantly monitored or supervised by staff, separate from goods likely to be purchased by persons under 18.	
Responsible promotions	That, if any advertising campaigns or promotions of alcohol are planned, the premises licence holder shall ensure they are carried out in a socially responsible manner and that they do not encourage irresponsible drinking. If authorised officers raise concerns about any promotion the premises licence holder shall discontinue it.	

The Southwark proof of age (SPA) card

- 252. In order to assist licensees avoid under-sales, the council's trading standards service has introduced a local PASS approved proof of age card. The SPA card is available to 16-18 year olds via their school or college. It is also made available through other youth settings by individual application up to the age of 25.
- 253. There are currently around 3,500 SPA cards in circulation and 7 of the 13 Southwark secondary schools and colleges have a process in place to make the card available to all students.
- 254. Anecdotal evidence is that many clubs will only accept driving licences and passports as they are concerned that other cards can be too easily copied. This is not the case. In fact the most commonly copied document is the driving licence. Cards bearing the PASS hologram cannot be easily copied.
- 255. PASS approved proof of age cards have recently been publicly supported by the Home Office and ACPO as reported in 'The Publican' http://www.thepublican.com/story.asp?sectioncode=7&storycode=68141&c=1 The Publican Wednesday 27th October 2010.
- 256. For further information on introducing an age verification policy, to access the training programme or to obtain the materials itemised above, please contact the trading standards service.
- 257. In detailing the council's recommended approach to under-age sales it is recognised that some sales may be made through intimidation or fear. Licensees experiencing such problems are advised to discuss their situation with licensing or trading standards officers. Furthermore, the council's health and safety team offers conflict resolution training aimed at helping deal with the threat of violence and aggression in the workplace. For contact details please see section 12.

Children and responsible drinks promotions

- 258. The Portman Group operates, on behalf of the alcohol industry, a code of practice on the naming, packaging and promotion of alcoholic drinks. The code seeks to ensure that drinks are packaged and promoted in a socially responsible manner and only to those who are 18 years and older. Complaints about products under the Code are considered by the Independent Complaints Panel and the Panel's decisions are published on the Portman Group's web site, in the trade press and in an annual report. If a product's packaging or point-of-sale advertising is found to be in breach of the code, the Portman Group may issue a retailer alert bulletin to notify retailers of the decision and ask them not to replenish stocks of any such product or to display such point-of-sale material, until the decision has been complied with.
- 259. This Code was first introduced in 1996 and has been developed to deal with issues ranging from "alco-pops" through to the marketing of alcoholic "energy" or "stimulation" drinks. It is an important weapon in protecting children from harm because it addresses the naming, marketing and promotion of alcohol products sold on licensed premises in a manner which may appeal to, or attract, minors.

260. The Council commends the Portman Group code of practice on the naming, packaging and promotion of alcoholic drinks to operators accordingly.

Children and cinemas

- 261. In the case of premises giving film exhibitions, the council will impose a mandatory condition that "access to any film exhibition will be restricted to only those who meet the required age limit in accordance with any certificate granted by the British Board of Film Classification, or in specific cases, the local authority". Applicants are expected to include within their operating schedule arrangements for restricting children from viewing age-restricted films.
- 262. Information regarding the classification of films by the council can be requested from the licensing service.

Children and regulated entertainments

- 263. This council acknowledges that Guidance states it may be necessary to impose a condition to restrict the admission of children to theatres which are incorporating adult entertainment in their productions.
- 264. Where a theatrical entertainment is specially presented for children, the council advises the presence of sufficient adult staff to control access and egress of children and to protect them from harm as follows
 - (1) An adult member of staff to be stationed in the vicinity of each of the exits from any level, subject to there being a minimum of one member of staff per 50 children or part thereof;
 - (2) No child (unless accompanied by an adult) to be permitted in the front row of any balcony; and
 - (3) No standing to be permitted in any part of the auditorium during the performance.
- 265. Where facilities generally are provided specifically for children, applicants are advised by the SSCB to consider the following steps:
 - That applicants should demonstrate that the company / organisation has a policy or statement on keeping children safe from harm;
 - That applicants should demonstrate how employees or contracted agencies will be made aware of the company's / organisation's policy / statement on keeping children safe from harm;
 - That applicants should demonstrate that recruitment processes for staff who
 may have to deal direct with children under the age of 18 seek to ensure, as
 best the company / organisation can, that the appointed applicant has not
 been known to have harmed children:
 - That applicants should demonstrate that there will be clear expectations on staff with regard to their personal conduct in direct relations with children under 18 and guidance on what is appropriate in dealing with specific situations which may arise;
 - That applicants should demonstrate that there are systems in place and that staff are aware of them for dealing with possible abuse or violence to children from other users of the licensed facility;

- That applicants should demonstrate that there are clear and accessible complaints systems for children to report if they have been harmed and that there are clear whistle-blowing procedures for staff who may have concerns about the welfare of children on the premises; and
- That all duty managers or staff in charge of the premises are aware of how to refer any concerns about the welfare of children to the local police, education authority or social services as necessary.
- 266. Where children are taking part in any regulated entertainment the council expects the arrangements set out in the council's "Guidance Note on Children in Entertainment" at the premises to assure their safety.

SECTION 11 - ENFORCEMENT

- 267. It is essential that licensed premises operate in accordance with the act. Part of the licensing authority's role is to monitor premises and take appropriate action to ensure that the licensing objectives are promoted. The licensing authority recognizes the importance of enforcement to the promotion of those objectives. The licensing authority is committed to continue to undertake enforcement measures and proceedings in partnership with other relevant statutory agencies. Offences against the act are set out in the Act.
- 268. In accordance with the guidance to the act, the licensing authority will work alongside the police and the other responsible authorities in promoting the licensing objectives. This licensing authority has an excellent relationship with each responsible authority and the police, in particular. Protocols exist with the local police on enforcement issues to provide for a more efficient deployment of local authority and police staff commonly engaged in enforcing licensing law and the inspection of licensed premises.
- 269. Protocols will also be maintained with the other enforcement bodies that have a role around the operation of licensed premises. As part of the "better regulation initiative" and in order to promote effective use of enforcement resources and reduce the burden on business, the various enforcement agencies involved may from time to time look at references normally dealt with by other relevant agencies during the course of premises inspections. Where issues are identified under those other agencies remits (for instance where a licensing officer notes a fire safety matter) this will be brought to the attention of the person in charge and the matter may be referred on to the agency involved. Where possible appropriate advice and guidance will be given at the time.
- 270. Inspections will be undertaken on a risk-assessment basis with resources targeted toward agreed problem and high-risk premises, which require greater attention. A lighter touch will be applied in respect of low risk premises, which are well run. Detail of the basis to our risk-assessment approach can be obtained from the licensing service.
- 271. This approach is supported and facilitated by regular liaison meetings of the primary responsible authorities and beyond through the partnership tasking group, public services operations group and local licensing enforcement group. All complaints and referrals made to such liaison meetings will be investigated appropriately.
- 272. All enforcement actions undertaken by the range of enforcement agents located within the council's environmental health and trading standards business unit (including the licensing service) will be in accordance with the council's enforcement policy; the Better Regulation Unit's Enforcement Concordat (to which this council is a signatory); and the Regulators Code of Compliance. As such it is our intention that all enforcement actions will be fair, transparent and proportionate. A copy of our enforcement policy may be obtained from the licensing service.
- 273. As part of this, it should be noted that investigations will be undertaken through a range of means including overt and covert operations. It should be noted, in

particular, that weights and measures officers from the council's trading standards service or officers from the police authority will carry out test purchases of alcohol using young volunteers, where complaints are received concerning the supply to children under the age of 18. Where sales are made all responsible persons may be subject to enforcement action.

- 274. Additionally, this authority recognises the important role played in preventing crime and disorder by the Security Industry Authority by ensuring that door supervisors are properly licensed and, in partnership with the police and other agencies, that security companies are not being used as fronts for serious and organised criminal activity. This authority will look to co-operate with the SIA as far as possible and as part of this co-operation has authorised officers to act under the powers of the Private Security Industry Act 2001.
- 275. It remains the over-riding intention, however, of the licensing authority, the police, and the other enforcement agents to work together with responsible licence holders and operators to help them run a successful business that plays a positive role within the local community. Where operators are acting responsibly we will look to provide support, advice and education and we will look to give early warning of any concerns identified at any premises. However, the authorities will use the range of enforcement powers made available under the Act where individuals act with disregard to their responsibilities.

SECTION 12 - CONTACT DETAILS

THE LICENSING SERVICE

Advice and Guidance

Further details of the licensing application processes and copies of application forms can be obtained from the licensing service. You can contact the service

In writing

The Licensing Service C/O Southwark Environmental Health & Trading Standards Business Unit The Chaplin Centre Thurlow Street London, SE17 2DG

By phone: Customer service centre on 020 7525 2000

By Fax - 020 7525 5705

By e-mail: <u>licensing@southwark.gov.uk</u>

By visiting the web site: www.southwark.gov.uk/businesscentre/licensing

OTHER RESPONSIBLE AUTHORITIES

Commissioner of Police for the Metropolis

The Police Licensing Office Walworth Road Police Station 12 – 28 Manor Place London, SE17 3RL Tel: 020 7232 6210

The London Fire and Emergency Planning Authority

Fire Safety Regulations (Southwark Borough) 249-259 Lewisham High Street London, SE13 6NH

Tel: 020 8555 1200 (x36500)

Fax: 020 8536 5924

The Environmental Protection Team (the "responsible authority" dealing with nuisance and pollution issues)

The Occupational Health and Safety Team The Trading Standards Service

C/O Southwark EHTS Business Unit

The Chaplin Centre Thurlow Street

London SE17 2DG

Tel: 020 7525 2000

The Council's Building & Development Control Services (including Planning)

Council Offices 5th Floor Hub 2 Tooley Street London, SE1

Tel: 020 7525 2000

The Local Safeguarding Children Board

Southwark Social Services C/O Council Offices 160 Tooley Street London, SE1

Tel: 020 7525 4646

Notes:

- If you are applying for a premises or a club premises certificate and part of the premises concerned falls within another licensing authority's area then that authority should be copied in on the application also.
- If the premises concerned is a vessel then a copy of the application should also be sent to the Navigation authority within the meaning of section 22(1) of the Water Resources Act 1991

OTHER AGENCIES

Patrick Horan
Access Officer &
Chair of Southwark Disablement Association

2 Bradenham Close London, SE17 2QB Tel: 020 7701 1391

Environmental Enforcement Team (providing the Council's 24 hour rapid response service)

C/O Southwark Community Safety Business Unit The Chaplin Centre Thurlow Street London, SE17 2DG

Tel: 020 7525 2000

Street Trading (for shop front licences)

Street trading office Dawes Street London, SE17

Southwark Events Team

C/O Council Offices 160 Tooley Street London, SE1